



Eidgenössische Technische Hochschule Zürich Swiss Federal Institute of Technology Zurich

Master Thesis

Chair of Logistics Management

A Study on Lean Implementation at a Humanitarian Organization

Author:

Maximilian Ebner

Am Wasser 48, 8049 Zürich

E-Mail: maebner@student.ethz.ch

Supervisors:

Professor Dr. Stephan Wagner Sarbani Bublu Thakur-Weigold

Zürich, August 2017

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng



http://www.ub.tuvŚwiśsi Federal Institute of Technology Zurich



The approved origination master thesis is available. Vienna University of **Declaration of Originality**

http://www.ub.tuvThe signed declaration of originality is a component of every semester paper, Bachelor's thesis, Master's thesis and any other degree paper undertaken during the course of studies, including the respective electronic versions.

> Lecturers may also require a declaration of originality for other written papers compiled for their courses.

I hereby confirm that I am the sole author of the written work here enclosed and that I have compiled it in my own words. Parts excepted are corrections of form and content by the supervisor.

Title of Master's thesis (in block letters):

A Study on Lean Implementation at a Humanitarian Organization

Authored by (in block letters):

For papers written by groups the names of all authors are required.

Name(s):	First name(s):
Ebner	Maximilian

With my signature I confirm that

- I have committed none of the forms of plagiarism described in the 'Citation etiquette' information sheet.
- I have documented all methods, data and processes truthfully.
- I have not manipulated any data.
- I have mentioned all persons who were significant facilitators of the work.

I am aware that the work may be screened electronically for plagiarism.

Place, date

Signature(s)

Zürich, 31.08.2017

For papers written by groups the names of all authors are required. Their signatures collectively guarantee the entire content of the written paper.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology.

Acknowledgements

http://www.ub.tu Foremost, I would like to thank Bublu Thakur-Weigold and Professor Stephan M. Wagner for supervising this thesis project. Special thanks to Mrs. Thakur-Weigold for putting in all the effort to make this project happen and never giving up even though things did not look as promising in the beginning. I want to thank Mrs. Thakur-Weigold also for her guidance during the project and for putting the trust in me to conduct this thesis.

I would like to show my greatest appreciation to Andrew Parris. I want to thank Mr. Parris for not only taking me in, but mentoring me during the whole project. I could always rely on his support which really helped me through the more difficult times of this project. Mr. Parris not only provided me his expertise on lean, but gave me the opportunity to work as his assistant during the lean roll-out at Medair and participate in the Green Belt training. I greatly benefited from Mr. Parris who took a lot of his time to guide me and provide detailed and constructive feedback on every section of this thesis. Without his help this thesis would not have been possible.

I want to thank Medair and all employees at Medair who welcomed me so well into the organization. Everyone treated so friendly which made me feel comfortable from the first day on. It was a pleasure to work at Medair and I will always look back at the time there fondly.

I am deeply grateful to my parents, Britta and Anton, and my sister Antonia, who always supported me at every stage of my life and especially during my studies. Their belief and trust in me give me the strength and endurance to face and overcome the most difficult obstacles and to strive for far and hard to reach goals.

Special thanks to my aunt and uncle, Renate and Fritz, who supported me so unconditionally during my stay in Switzerland. Thank you for always being there for me.

I want to thank all of my friends who always supported me during my studies. Thank you for all the support and the good times we shared.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Tech-nischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at		
master thesis is available at Vienna University of Technolo Abbrey http://www.ub.tuwien.ac.at/	he main library of the gy. Jations	iv
List of	Figures	v
List of		vi
1 A	ostract	
2 In	troduction	_ 7 _
2.1	Efficiency for Humanitarian NGOs	
2.2	Introduction to Lean	
2.	2.1 Definition of Lean	4 -
2.	2.2 The Evolvement of Lean Understanding	5 -
2.3	Lean Applied Outside Manufacturing Companies	7 -
2.4	Motivation for Study	8 -
3 14	an Annlicability Assessment	- 10 -
3 1	literature Review	- 10 -
3.1	1.1 NGO Context	- 10 -
3.	1.2 Difference between Development and Humanitarian Organizations	- 12 -
3.	1.3 Specific Humanitarian Context (Van Wassenhove, 2006)	14 -
3.	1.4 Growing Importance of Humanitarian NGOs in Disaster Relief.	15 -
3.2	The Differences between the Humanitarian Aid and Private Sector	15 -
3.3	Assessment of Contextual Differences in Terms of Lean Applicability	21 -
3.	3.1 Literature Review	21 -
3.	3.2 Methodology of Assessment	22 -
3.	3.3 Assessment	23 -
3.	3.4 Discussion	27 -
3.	3.5 Conclusion	28 -
4 Po	otential Benefit and Need of Implementing Lean for Humanitarian NGOs	30 -
4.1	Lean Loois for NGOs	30 -
4.	1.1 Literature Review of Existing Implementations	
4.2	Lean Thinking for Humanitarian NGOS	
4.	2.2 Loop Loodorshin	- 30 -
4.	Need of Lean for Humanitarian NGOs	- 11 -
4.3	A Lean and the Core Humanitarian Standard	- 12 -
4.4	Conclusion	45 -
5 Ca	ise Study: Medair	46 -
5.1	Overview of Organization Under Study	46 -
5.	1.1 Vision & Mission	46 -
5.	1.2 Growth & Financials (Medair, 2016)	47 -
5.	1.3 Working Areas and Organizational Structure	49 -
5.2	Unique Humanitarian Context – Medair	52 -
5.	2.1 General Challenges for Medair	53 -
5.	2.2 Faith-Based Organization and Lean	55 -
6 N	edair`s Lean Roadmap	57 -

10 01		
1.0.00		. 11
	4.	

Die approbierte Originalversion dieser Diplom/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbiblio

The approved origin	6.1	Process Excellence Manager	- 57 -
naster thesis is avail	lable 6.11.1	^{nain} The Pérson	- 57 -
nttp://www.ub.tuwie	en.ac.a(.1.2	The Role	- 58 -
	6.2	The Lean Roll-Out	- 59 -
	6.2.1	Process Excellence Energizer Training (Parris, Process Excellence Energizer, 2017)	- 60 -
	6.2.2	Lean Six Sigma Green Belt Training	- 60 -
	6.2.3	Building Lean Leaders	- 61 -
	6.2.4	Deploy Lean to the Field	- 61 -
7	Revi	ew of Medair`s Approach	63 -
	7.1	Stage 1: Process Excellence Energizer Training	- 63 -
	7.2	Stage 2: Developing Change Agents and Lean Leaders	- 64 -
	7.3	Stage 3: Improvement Project	- 65 -
	7.4	Stage 4: Lean Daily Meetings and Visual Boards	- 66 -
	7.5	Bringing Lean to the Field	- 66 -
8	Con	clusions	67 -
	8.1	Answer Research Question 1: Can Lean be applied to benefit a humanitarian NGO?	- 67 -
	8.2	Answer Research Question 2: If so, what aspects of Lean are most useful?	- 67 -
	8.3	Answer Research Question 3: How has it been deployed in Medair and other humanitar	rian
	NGOs?	- 68 -	
	8.4	Limitations	- 69 -
	8.5	Recommendations for Medair	- 70 -
А	ppendi	<i></i>	A
W	/orks Ci	ted	B

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

WIEN UB Universitätsbibliothek

The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng

Abbreviations

E	3PR	Business Process Reengineering
(CCBRT	Comprehensive Community Based Rehabilitation in Tanzania
(CEO	Chief Executive Officer
(CHS	Core Humanitarian Standard
(CI	Continuous Improvement
(CIP	Continuous Improvement Process
[DMAIC	Define-Measure-Analyze-Implement-Control
E	ELT	Executive Leadership Team
C	GDP	Gross Domestic Product per capita
ŀ	HQ	Headquarter
ŀ	HQAI	Humanitarian Quality Assurance Initiative
I	MVP	International Motor Vehicle Program
I	NGO	International non-governmental organization
l.	SO	International Organization for Standardization
ľ	Т	Information Technology
J	JIT	Just-In-time
ŀ	<pi< td=""><td>Key Performance Indicator</td></pi<>	Key Performance Indicator
L	_AI	Lean Aircraft Initiative
Ν	MIT	Massachusetts Institute of Technology
Ν	MSF	Medicines Sans Frontier
١	NGO	Non-Governmental Organization
١	NIST	National Institute of Standards and Technology
(DECD	Organization for Economic Co-operation and Development
(DEE	Overall Equipment Effectiveness
F	PDCA	Plan-Do-Check-Act
F	PEM	Process Excellence Manager
F	PMO	Project Management Office
5	SMED	Single Minute Exchange of Die
T	ГРМ	Total Productive Maintenance
٦	ГQM	Total Quality Management
٦	TSO	Third Sector Organization
ι	JN	United Nations
V	NIMB	We-I-Made-it-Better
V	NVI	World Vision International

TŲ UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.aEfforture 3-1. Dise

List of Figures

ac	Figure 3-1: Disaster Management Cycle adapted from Van Wassenhove (2006, p	p.
	481) 1	12 -
	Figure 3-2: Humanitarian Space (Van Wassenhove, 2006, p.478) 1	14 -
	Figure 3-3: Customer-Buyer Relationship in Both Contexts 1	18 -
	Figure 4-1: Business Principles of Toyota (Liker J., The Toyota Way: 14 Management Principles from the World's Greatest Manufacturer, 2003)	38 -
	Figure 4-2: CHS Standards on Quality and Accountability (Core Humanitarian Standard, 2014, p. 4)	13 -
	Figure 4-3: Process Excellence 4	13 -
	Figure 5-1: International Humanitarian Response, 2011-2015 (Development Initiatives, 2016, p.36) 4	18 -
	Figure 5-2: Operational Expense Medair 2016 4	19 -
	Figure 5-3: Medair Overview (Medair, 2016) 4	19 -
	Figure 5-4: Organizational Chart Medair (July 2017) 5	51 -
	Figure 6-1: Medair`s Lean Roll-out 5	59 -
	Figure 8-1: Possible Cross-functional Structure at Medair	72 -

TU UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbibliothek

The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.areable 2-1. Num

List of Tables

.ac	Table 2-1: Number of LinkedIn Groups for Improvement Methodologies (Baudin,	
	2013 & LinkedIn, 2017) 3	-
	Table 3-1: Distinction between Different Sectors (adapted from Abad, 2013) 11	-
	Table 3-2: Score System 23	-
	Table 3-3: Lean Applicability Assessment 24	-
	Table 4-1: Lean Tools and Their Applicability for Humanitarian NGOs (adapted from Chandra, 2013) - 31	-
	Table 4-2: Different Leadership Traits (Parris, Thoughts on Lean in HumanitarianAid, 2017)	-
	Table 4-3: Process Excellence Match with CHS (Parris, Thoughts on Lean inHumanitarian Aid, 2017)- 44	-
	Table 5-1: Medair Programmes (Medair, 2016) 50	-
	Table 5-2: Medair Characteristics 53	-
	Table 7-1: I-We Made it Better (WIMB)-Improvements 64	-
	Table 7-2: Green Belt Training Participants 65	-

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng



This study assesses the applicability of lean for humanitarian aid organizations and what of the lean methodology is useful for humanitarian organizations. A 15 billion USD shortfall in meeting all humanitarian needs motivates humanitarian aid non-governmental organizations (NGO) to improve their efficiency and performance. Lean can reduce the impact of this financial gap and enable humanitarian NGOs to relieve more suffering.

An assessment based on existing literature and a lean expert's experience investigates how the differences in context between business and humanitarian NGOs affect the applicability of lean for humanitarian work. It shows that humanitarian aid organizations face more challenging obstacles than business organizations to become lean due to the contextual work differences.

Nevertheless, this work asserts that there is an urgent need for lean in the humanitarian aid sector because it provides tools and an organizational culture that can achieve more with the limited funds they receive. The assessment of the applicability of lean tools to humanitarian operations shows that most lean tools are applicable for humanitarian operations. The lean organizational culture can benefit humanitarian organizations because lean thinking is an organizational mindset of continuous improvement which is not specific to any work context.

This thesis also reviews the first organization-wide lean implementation at Medair, an international humanitarian NGO. This case study shows how lean can be adopted to the context and how this lean transformation is being rolled out in a humanitarian NGO.



TU UB

The approved original v2orintroduction master thesis is available 2 thintroduction Vienna University of Technology.

ttp://www.ub.tuwien.ac.at/enc

Every day we see disturbing images of humanitarian situations on the news: A Syrian mother with her six children that lost her home, now living in poor conditions, in a refugee camp, one of her children has lost one or more extremities, most of them are sick. On the next report, we see a United Nations truck, arriving in a South Sudanese camp, to deliver food. People are jumping on the truck snapping as much food as possible, two aid workers on the truck trying to re-establish order, but failing. Tragedies and situations like this are part of a humanitarian aid worker's everyday life.

Humanitarian operations save lives, alleviate suffering, and maintain human dignity following conflicts, shocks, and natural disasters (OECD, 2016). Humanitarian non-governmental organizations operate in these unstable and unpredictable environments. According to the United Nations (UN), only 62% of the identified humanitarian need has been met, in 2014. This is the biggest shortfall ever, although the raised funding has increased over tenfold over the last 15 years, from 2 billion in 2000 to 24.5 billion USD in 2014. The financial gap to meet all humanitarian needs is estimated to be 15 billion USD (Development Initiatives, 2016, p. 24). To close this gap, there are two possible solutions:

- 1. Donors provide more funds
- 2. Humanitarian operations become more efficient

Both solutions must be applied together, to be able to meet all humanitarian needs (Parris, Thoughts on Lean in Humanitarian Aid, 2017). This work studies the efficiency solution to close the gap, by implementing a process improvement methodology, called lean, in a humanitarian NGO.

2.1 Efficiency for Humanitarian NGOs

For humanitarian agencies there are two motivations to become more efficient. Internal motivation, so that the organization is able to relieve more people from suffering, and an external motivation, to satisfy donors. Donors demand increasing levels of efficiency and effectiveness. Effectiveness measures the extent to which the customer needs are met, while efficiency measures how economically resources are utilized to a given a level of

The approved orginal determination of the services, and drinking water – are met by humanitarian operations; efficiency

measures how many resources, e.g. funds, are needed to reach a level of effectiveness. Since resources are limited for all humanitarian organizations, these organizations have a strong incentive to become more efficient.

Many models and methods have been developed to increase efficiency and performance in the business sector. Business process reengineering (BPR), total quality management (TQM), lean, and six sigma are well-known and widely applied models for improving processes, operations, and the organization's overall performance (Näslund, 2008, p. 271f). All models have some characteristics in common, such as continuous improvement, customer focus, people, and management involvement (Näslund, 2008, p. 273). Nevertheless, the models present important differences due to their different origins and history (Chiarini, 2011, p. 332). Among the several quality management tools, lean is one of the more successful and wide-spread methodologies (Andersson, Eriksson, & Torstensson, 2006, p. 288). Lean and six sigma are by far the most popular methodologies today. To prove this, Baudin (2013) assessed the number of groups on LinkedIn that are found, when searching for the methodology's name, the results are shown in Table 2-1. To show the trend from 2013, the same is done in June 2017 and added to Table 2-1. Some of the groups are overlapping, but this table still draws a clear picture of the improvements methodology's popularity.

Improvement Methodology	LinkedIn Groups (Baudin, 2013)	LinkedIn Groups 2017
Lean	2.642	2.796
Six sigma	1.411	1.010
TQM	87	63
BPR	16	44

Table 2-1: Number of LinkedIn Groups for Improvement Methodologies (Baudin, 2013 & LinkedIn, 2017)

Baudin (2013) argues that TQM and BPR are "dead", and that the only methodology that thrives is lean. This seems to be confirmed, given the numbers of LinkedIn groups in 2017, where there is a clear decline of groups in six sigma. Furthermore, Baudin (2013) states that the LinkedIn group numbers only make a statement about the popularity of the

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

The approved original methodologies and not the technical effectiveness of the approaches, but the numbers master thesis is available at the main library of the Vienna University indicate that lean has by far the highest market share and is still growing. Sections 2.2 and 2.3 aim to first understand what lean is and then to explain what makes the lean concept so successful and broadly applicable.

2.2 Introduction to Lean

Lean, along with its variant Lean Six Sigma, has become a global management movement in many industries (Liker, 2016, p.9). Processes and all stakeholders of processes such as employees, suppliers, and customers are the topic of lean. The aim of lean is to continuously improve processes towards a state of perfection, the one-piece flow.

Lean was first mentioned by John Krafcik in 1988 (Netland & Powell, 2017, p.xxi), but made popular by the famous book *The Machine That Changed the World* in 1990, which derived from a five-year automobile industry study conducted by the International Motor Vehicle Program (IMVP), a programme of the Massachusetts Institute of Technology (MIT) (Liker, 2016, p.13). The result of this study show that Toyota, other Japanese automobile manufacturers, and their suppliers outperform American and European competitors by managing organizations differently.

The Machine That Changed the World not only revealed Toyota's outstanding performance, but also tried to study and explain what made Toyota unique. This book provided the impetus for Western manufacturing companies to enter into a new era of production and management practices. And now, after 27 years, it is considered best practice in the manufacturing industry and has been spread into various other sectors such as start-ups, services, health, and construction. (Netland & Powell, 2017, p.xxi)

2.2.1 Definition of Lean

After the introduction of lean in 1990, many researchers, consultants and professors have studied, developed, and taught various versions of lean according to their own understanding (Liker, 2016, p.9). This leads to a broad spectrum of applications, but makes it difficult to find consensus among lean experts about what lean really is. Various

The approved original vectors of the main library of the Vienna University (Goldensohn, 2015). To better understand lean, it is helpful to look back at origin of this http://www.ub.tuwien.ac.at/eng term used in a management context.

John Krafcik, a graduate student who took part in the IMVP, argued that lean means doing more with less (Liker, 2016, p.13). Toyota needed less time, space, employees, and money to develop, produce, and sell automobiles than their non-Japanese competitors. Not only did Toyota use fewer inputs, they also achieved better results in terms of product quality, higher service level, and innovations than non-Japanese car manufacturers (Womack, et al., 1990, p.118).

Another definition from Paul A. Akers (2016) which does not focus simply on efficiency, states: "Lean is about eliminating and about continuously improving" (p.26). The first part of this definition focuses similar to Krafcik's definition on efficiency, but Akers also highlights continuous improvement, which is established through an organizational culture. The lean culture is a necessity for becoming a lean organization, which was not understood in the early years of studying Toyota (see Section 2.2.2).

The United States National Institute of Standards and Technology (NIST) defines lean as the following: "A systematic approach to identifying and eliminating waste through continuous improvement, flowing the product at the pull of the customer in pursuit of perfection." (NIST, 2000) This definition completes the two definitions above, because it also states the importance of flow and pursuit of perfection, which are two lean fundamentals.

2.2.2 The Evolvement of Lean Understanding

Lean has significantly evolved since the late 1980s. At first, it was considered a technical approach for manufacturing enterprises to improve efficiency, based on applying Toyota's techniques such as Just-in-Time (JIT) and Kanban (Holmemo, Rolfsen, & Ingvaldsen, 2016, p. 2). To be able to increase efficiency, a company has to reduce waste. Taiichi Ohno (1998), widely-acknowledged as the father of lean manufacturing (Smith & Hawkins, 2004, p. 7), defines waste as all activity that add costs to the product but not value. The following are his seven types of waste:

 III
 UB

 Die approbierte Originalversion dieser Diplom-/

 Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

nischen Universität Wien aufgestellt und zugänglich http://www.ub.tuwien.ac.at

- Introduction -

TU UB WIEN Universitätsbibliot

3.

The approved original version of this diponent of the version version of the version version of the version version version of the version v

Motion: Movement of people that does not add value

4. Waiting: Idle time created when material, information, people, or equipment

is not ready

5. Processing: Efforts that adds not value from a customer's perspective

6. Inventory: More material, parts, or products on hand than the customer needs right now

7. Defects: Work that contains errors, rework, mistakes, or lacks something necessary

Womack and Jones (1996) introduced an additional type of waste:

8. Unused Employee Genius: Underutilizing capabilities, delegating tasks with inadequate training

Lean offers tools to eliminate all of these types of waste. A description of 13 imporant lean tools can be found in Section 4.1. Many people mistake lean for only the tools the methodology offers.

Dombrowski & Mielke (2013) did a study on 91 enterprises in industries, mainly in the German automotive sector. Their analysis showed that 92% of the enterprises use the broadly applied lean tools: 5S and continuous improvement process (CIP). 80% say they use lean principles, but only 5% of German participants say that employees live the core values of the enterprises. An improvement culture is hard to determine, but still the company's values have to be lived by employees and leaders, otherwise an improvement culture will not be successful (Dombrowski & Mielke, 2013, p. 572). Rother (2012) argues that only 1% of companies have successfully implemented a lean culture. This shows the important gap between only applying lean tools and implementing a lean organizational culture of continuous improvement. The improvement culture is an important cornerstone, that is often neglected.

n.ac.at

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TŲ UB

Around 2000s, Jean was understood as a management system for the whole was understood as a management system for the whole on organizational learning and contingencies. Researchers focused on the gap between Toyota and Western companies, which tried to implement lean. Different gaps were identified and books on lean leadership (Liker & Convis, The Toyota Way to Lean Leadership, 2011), lean culture (Liker & Hoseus, Toyota Culture: The Heart and Soul of the Toyota Way, 2008), and lean coaching (Rother, 2009) were published. Furthermore, lean has been described as a general 'philosophy' that can be adapted and translated to different business contexts. (Holmemo, Rolfsen, & Ingvaldsen, 2016, p. 2)

> Mike Rother (2009) argues in his book *Toyota Kata* that establishing a lean culture can only be achieved by building capabilities in people through coaching. A lean organizational culture is one where the primary goal is to continuously improve (Rother, 2009, p. 23ff). This work assesses basic conditions that need to be met in order to implement lean. Section 3.3 presents the obstacles that a humanitarian NGO has to overcome to develop a lean culture.

2.3 Lean Applied Outside Manufacturing Companies

Lean has achieved great results for manufacturing companies, word spread out and companies in other sectors wanted to benefit from this methodology too. Netland and Powell (2017) published the book *The Routledge Companion to Lean Management*, a book that's contains of lean application outside manufacturing industries, for instance applications in marketing, IT, education, and even police. This paper will show that lean offers a way of thinking and a tool set to improve performance that can be applied to the humanitarian aid sector as well. This broad applicability makes lean so powerful.

The focus of lean – process improvement – can be applied to areas outside manufacturing, because all work occurs in processes (Parris, Green Belt Training, 2008). Netland (2016) explains why lean is so broadly applicable, in the introduction of *Lean Management Journal*, by stating: "Wherever there is a process, lean can be used to increase efficiency." (p.9) This means that lean can help every organization. To better understand what process means, Parris (2008) provides a definition: "A process is a repeated sequence of actions that transforms inputs into outputs and leads to a value

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TŲ UB

The approved original version of the customer." This definition already provides the customer focus, which is master thesis is available at the main library of the Vienna University important in lean.

Lean represents an industrial paradigm shift from an internally-focused operations orientation to a customer-focused approach (Bowen & Youngdahl, 1998, p. 215). In lean processes, steps are identified as value adding or non-value adding based on the perspective of the customer. The purpose of a process is to serve a customer. The lean methodology of identifying and eliminating waste by improving continuously is used to improve an organization's performance. One of the most powerful lean tools called value stream or process mapping, shows that lean can be applied outside its original context. Section 4.1 assesses lean tools and the extent to which they can benefit humanitarian organizations.

2.4 Motivation for Study

As the introduction shows, lean has been applied in various areas outside its origin, the Japanese automobile manufacturing industry. Humanitarian organizations are under tremendous pressure to operate efficiently and to improve their processes. NGOs, especially humanitarian NGOs, are usually 15 years behind their counterparts in business (Van Wassenhove, 2006, p. 476). In line with this, lean has hardly been applied in the humanitarian aid context, one notable exception, World Vision International (WVI), an international NGO, where lean showed significant success: reducing the annual costs by \$ 1.5 million and reducing the targeted span times by 59% (Parris, Fighting Poverty in East Africa with Lean Six Sigma, 2015). Still, WVI did not deploy lean globally, but primarily in its East Africa countries.

Considering the different contexts of the manufacturing industry, from where lean originates, and humanitarian aid, it has to be assessed if lean is able to benefit humanitarian aid organizations. This leads to the first research question:

Research Question 1:

1. Can lean be applied to benefit a humanitarian NGO?

Die approbierte Originalversion dieser Diplom// Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

TŲ UB

TU UB

- Introduction -

The approved original To answer this question, this paper assesses the impediments humanitarian NGOs face master thesis is available at the main library of the Vienna University in order to become http://www.ub.twien.ac.at/eng the for-profit sector and the humanitarian aid NGO world impact the ease and extent of applicability of lean.

Research Question 2:

2. If so, what aspects of lean are most useful?

Is it beneficial to implement an organizational culture of continuous improvement or better to only apply lean tools that make processes more efficient? Section 4.1 assesses how lean tools can benefit humanitarian operations and a literature review shows how lean tools have already been applied to improve operations. Section 4.2 shows that lean thinking and implementing a lean culture can benefit humanitarian NGOs.

Research Question 3:

3. How has lean been deployed in Medair and other humanitarian NGOs?

There are some lean implementations in humanitarian aid already which this thesis will show. Furthermore, this work studies how a lean initiative has been rolled out recently at Medair, an international humanitarian NGO based in Switzerland. Given the novelty of an organization-wide lean implementation for a humanitarian NGO, this is an interesting topic for academic research. The case study describes how lean is being rolled out at Medair, how successful the first five months of the initiative were, and gives recommendations to Medair and other humanitarian NGOs that want to implement lean.



TU UB

The approved original v3 or Location Applicability Assessment master thesis is available 3 th Lean Applicability Assessment Vienna University of Technology.

To study if lean can benefit a humanitarian NGO, it is necessary to understand the context in which non-governmental humanitarian aid organizations work. In this chapter, a literature review provides the different contexts to business and defines the specific context for humanitarian NGOs. Later in this section, ten fundamental differences between the business context and that of a humanitarian NGO are identified and discussed. This is done in order to investigate the influence of these differences on the extent to which lean can be applied in the humanitarian aid sector. Some of these differences may impede, enhance, limit, or do not affect the applicability of lean.

3.1 Literature Review

3.1.1 NGO Context

To understand humanitarian organizations better, one has to look closer into the specific context, a humanitarian international non-governmental organization, operates in. There are significant differences between international NGOs and the corporate business world, but even further differences, when it comes to the specific humanitarian aid context.

Before analyzing the differences in contexts, it is worth taking a closer look at the history of NGOs. National NGOs and international NGOs are legally the same organization and have similar contexts, except for the national or international focus of their work. Although organizations which worked in similar contexts as NGOs had been active since the eighteenth century, the term NGO is a post-World War II term, was introduced by the United Nations (UN) Charter Article 71, in 1945 (Lewis, 2009, p. 1057). The English Oxford Dictionary (2017) defines NGO as following: "A non-profit organization that operates independently of any government, typically one whose purpose is to address a social or political issue." One fundamental difference between corporate businesses and NGOs is the aim of achieving goals without pursuing profit. Non-profit organization is a frequently used term, in the US, referring to a NGO (Lewis, 2009, p. 1052), because for NGOs it is forbidden to make profit. Today, the acronym NGO tends to be used to relate to international or development country work (Lewis, 2009, p. 1052). Due to the origin of Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

TU UB

- Lean Applicability Assessment -

The approved original the term in library of the UN Charter, where the designation for NGOs was awarded to Vienna University international non-state organization that were given consultative status in UN activities http://www.ub.tuwien.ac.at/eng (Lewis, 2009, p. 1052).

Although NGOs are forbidden to make profit, funds are vital and usually accumulated through donations or membership fees. Independence from public or private entities is an important characteristic of NGOs, which is ensured with a certain degree of financial autonomy. NGOs often fill gaps in government services or help certain population groups, therefore a healthy relationship with the government is critical. Although depending too much on a single source, especially a public source, endangers the organization's independence and might influence the own mission of an NGO towards being or replacing a public service. In this situation, a NGO becomes mainstream and rather than an alternative. (Clark, 1995, p.596)

Clearly differentiating from private and the public sector, there is a third sector, the societal sector, including NGOs, non-profits and voluntary organizations, referred to as *third sector organizations (TSO)* (Billis, 2010, p.4). Inside this sector, organizations vary strongly in size and operating field. From a local charity organization, providing books to students to a global billion dollar NGO, such as Amnesty International. Table 3-1 shows the differences between the private, public, and third sector organizations.

	Private	Public	TSO
Sphere of Operation	Market	State	Local or international Communities
Purpose	Profit	Public service	Public good
Fund Generation	Profit making	Taxation	Donations, Membership fees
To whom accountable	Owners	Public through government officials	Donors and beneficiaries
Client Participation	Through buying of goods or services	Originally as passive recipients	Through receiving of goods and services

Table 3-1: Distinction	hetween	Different	Sectors	(adapted from	Abad	2013)
	Detween	Different	000000	lagapted nom	Abdu,	2010)

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TUUB

- Lean Applicability Assessment -

approved origin 3.1.2 of this d Difference between Development and Humanitarian r thesis is available at the main library of the a University Organizations

Inside the third sector, the NGO sector, some organizations operate in the humanitarian and development sector. To clearly define what a humanitarian NGO is and does, this section shows the differences between humanitarian aid and development NGOs, which are commonly confused. For both, work usually occurs in developing countries, because the local government is not able to provide enough resources to overcome weaknesses. The initial cause for need of humanitarian aid is usually a man-made or natural disaster, for instance, wars, severe corruption, droughts, outbreak of diseases, or earthquakes. Looking at a four phase model of disaster management from Van Wassenhove (2006) it becomes clear, where the differences between humanitarian aid and development work lie.



Figure 3-1: Disaster Management Cycle adapted from Van Wassenhove (2006, p. 481)

According to Warfield (2008), humanitarian agencies are called upon to deal with immediate response (stage 3) after a disaster hit. The aim of the emergency response it to maintain life, improve health and provide assistance to relieve human suffering and ensure people's dignity (OECD, 2016). Being prepared as an organization and having stock in the right places to be able to respond rapidly, is also part of the humanitarian area (stage 2). As the emergency is brought under control, the recovery phase (stage 4) begins. There is no clear distinction point where the emergency response turns into

TU UB

- Lean Applicability Assessment -

The approved original extended in the main library of the Vienna University development and humanitarian work, for instance building earthquake resistant houses in endangered by earthquake zones. (Bray, 2001, p.77)

In unresolved protracted crisis, the need of humanitarian aid can last for many years. The relationship between humanitarian response and development aid has been a broadly-debated issue (DARA, 2013, p. 7). Some argue that humanitarian aid has to pave the way for development aid, others say development work stands alone (DARA, 2013, p.7). Problems occur when response programmes transit into development programmes and usually different agencies take on the projects. The nature of projects are different, development projects are usually long-term and have a clear future goal, while humanitarian aid focuses on saving lives immediately. Humanitarian and development NGOs have therefore developed different cultures, thinking, and views on projects.

Ferris (2007) addresses four particular gaps, between the humanitarian and development world:

1. Institutional gaps: Different operational styles and cultures exist between different organizations and governments.

2. Financial gaps: Funding is often either for emergency or development assistance.

3. Temporal gaps: A particular gap emerges after emergency starts to subside and before long-term development work begins.

4. Programme and Funding cycle gaps: Usually development agencies use multi-year planning cycles, while humanitarian organizations use shorter time perspectives.

Looking at these four differences it becomes obvious that different understandings and organizational cultures between the different fields exist. Development work is challenging, because it naturally occurs in underdeveloped regions with low levels of education, security, infrastructure, and health. In situations the local government is not able to develop regions by its own resources, development NGOs provide assistance. Humanitarian aid can be seen as even more challenging than development work,

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

FU UB

- Lean Applicability Assessment -

The approved original decause, the work occurs in the same areas as development work, but right after a matter thesis is available at the main library of the veloping disasters occur in developing and developed countries, but hit developing countries harder due that infrastructure is poorer and governments are overwhelmed, thus NGOs are needed to help. Humanitarian aid is necessary not only after natural disaster, but also during and after man-made crisis, for instance conflicts. Considering these circumstances, humanitarian aid work is psychologically and physically demanding. Burnouts rates are high in the humanitarian aid sector (Eriksson, et al., 2009, p. 671).

3.1.3 Specific Humanitarian Context (Van Wassenhove, 2006)

Figure 3-2 provides a visualization the humanitarian space as dynamic triangular of the three principles of humanitarian organization: humanity, neutrality, and impartiality. The principles of humanity, impartiality, and neutrality are derived from the Fundamental Principles of the International Red Cross and Red Crescent Movement proclaimed in Vienna in 1965 by the 20th International Conference of the Red Cross and Red Crescent; United Nations (UN) General Assembly Resolution 46/182, 19 December 1991; and UN General Assembly Resolution 58/114, 5 February 2004. Humanitarian NGOs are expected to help anyone in need, regardless their religion, nationality, or other factors; not influence the outcome of a conflict; and not favor one group over the other. These three elements define the context in which humanitarian organizations operate.



Figure 3-2: Humanitarian Space (Van Wassenhove, 2006, p.478)

This space should define a physical zone of tranquility where aid workers, civilians, and non-combatants can move and operate freely. Under international conventions, political and military actors are prohibited to encroaching this space, or impeding aid work, but in

TU UB

- Lean Applicability Assessment -

The approved original validation of the validati

3.1.4 Growing Importance of Humanitarian NGOs in Disaster Relief

The importance of NGOs in responding to disasters has been expanding for some time, although the public opinion is still that national governments and UN agencies lead international disaster responses (Gannon, 2014, p. 2). There are several reasons for this change (Osa, 2012, p. 67):

1. Disasters overwhelm the capacity of national governments, so NGOs are needed to relieve human suffering.

2. NGOs have unique capacities and functions that governments, UN agencies and business lack. NGOs:

- a. Mobilize funds and workers in a way a government cannot
- b. Help to connect various actors to one another
- c. Provide specialized services that communities need to rebuild

Time is the key element in emergency situations (Eriksson, et al., 2009, p. 671) which makes NGOs necessary, because they can mobilize faster.

3.2 The Differences between the Humanitarian Aid and Private Sector

After focusing how humanitarian NGOs work and why they are important, it is now necessary to investigate the differences between the humanitarian aid and the private sector, in order to assess the ease and extent of applicability of lean in humanitarian NGOs, from where lean originates. This section lists the ten most fundamental differences in context. The assessment in Section 3.3.3 will show how these differences influence the applicability of lean.

1. Purpose: Profit may not be the main purpose of commercial organizations, but it is substantial for any kind of business. It ensures the future of the company and provides funds for further investment and income for shareholders. It is the Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

- Lean Applicability Assessment -

TU UB WIEN Universitätsbibli

The approved original version financial reward for running a business. Humanitarian NGOs do not have this Vienna University of Technolog financial incentive because they are forbidden to make profits (Lewis, 2009, p. 1052). The purpose of humanitarian organizations is to assist people in need, save lives, and relieve human suffering.

> 2. Staff Availability & Turnover: Unlike the commercial sector where companies try to avoid risk by operating in safe and politically stable environments, humanitarian operations run in challenging to even hostile environments and not all potentially skilled people are willing to work in these dangerous places (Parris, Thoughts on Lean in Humanitarian Aid, 2017). Due to limited financial resources NGOs cannot attract top managers or consultants, which puts them out of reach of best practices in industry (Thakur-Weigold, Stumpf, & Wagner, 2015, p. 25). The working environment and financial limitations lead to a scarce pool of qualified and readily deployable personnel for the humanitarian NGO sector (Tomasini & Van Wassenhove, 2009, p. 551). Demanding work conditions in disaster and conflict areas, long separation from family and friends, limited career perspectives, low job security, and common notion of humanitarian aid as 'volunteering' endeavor contribute to a high number of employee-initiated job exits in the humanitarian aid sector (Henry, 2004, p. 24f). Short funding cycles make it difficult for organizations to offer their employees reliable career perspectives (Korff, Balbo, Mills, Heyse, & Wittek, 2015, p. 524). Therefore, retaining staff is more challenging and staff turnover is much higher compared to the commercial sector. High staff turnover has a negative effect on performance (Tomasini & Van Wassenhove, 2009, p. 551) and leads to higher costs. Costs occur directly in form of separation, recruitment, and induction costs; indirectly in loss of institutional memory and lack of handover (Loquercio, Hammersley, & Emmens, 2006, p. 10).

> 3. Motivation: Generally, the income level for workers in NGOs is lower compared to the commercial sector (Parris, Thoughts on Lean in Humanitarian Aid, 2017) which reduces the extrinsic motivation for employees to work for NGOs. In addition, financial incentives to work harder, such as paid overtime or end of year bonuses in commercial organization, rarely exist in humanitarian NGOs (Parris, Thoughts on Lean in Humanitarian Aid, 2017). In contrast, the work in

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

- Lean Applicability Assessment -

TU UB

The approved original version humanitarian aid may be motivated by an intrinsic drive to honor inner values and master thesk is available at the main library of the Viena University of Technolos fulfil basic psychological needs (McCormack & Joseph, 2013, p. 149). A recent http://www.ub.tuwien.ac.at/org study shows that, unlike the commercial sector, most of the external factors and some personal factors do not play a big role in humanitarian NGOs; internal factors and personal readiness of the person for humanitarian work are most critical (Dubey, Gunasekaran, Altay, Childe, & Papadopoulos, 2016, p. 212). A study by Legnerova (2016), confirms Drucker (1996), who argues that employees in the forprofit sector are mostly motivated by monetary rewards, while non-profit workers prefer a meaningful sense of their work (Legnerova, 2016, p. 338). Humanitarian NGOs have to look carefully at the personal fit for the person and try to create an environment where employees' intrinsic motivation can thrive.

4. Customers: To explain this difference, first the term customer need to be defined. This works takes the definition from Business Dictionary:

Customer: "A party that receives or consumes products (goods or services) and has the ability to choose between different products and suppliers." (WebFinance, 2017)

In the business context, companies usually have suppliers and customers as their main stakeholders or they only provide services and therefore have few suppliers. The customer pays in order to receive a product or service and is therefore both the buyer and the recipient of the product or service. In the humanitarian NGO context, the customer, called beneficiary, of the product or service is not the buyer.

Institutional or private donors "buy" services of humanitarian NGOs so that someone else receives a benefit (Parris, Thoughts on Lean in Humanitarian Aid, 2017), as shown in Figure 3-3. In business transactions, the buyer receives the service or product and thereby gets information of the quality and can decide how well the service or product meets her or his expectations, and decide to buy again or not. In the NGO context the buyer does not automatically receive information about the product or service, and usually does not get direct feedback from the beneficiaries who receive their donations.

TŲ UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbiblio

- Lean Applicability Assessment -



To get information, donors demand a high level of transparency of NGOs and reporting, to assure that the money is spent to reach the beneficiary in the optimal way (Parris, Thoughts on Lean in Humanitarian Aid, 2017). A humanitarian NGO has to satisfy the two customers – the donor and the beneficiary – to be successful, while the commercial sector only has to satisfy a single customer. Donors are usually more powerful than those affected by disasters which may make the beneficiaries' requirements and perspectives of secondary importance (Oloruntoba & Gray, 2009, p. 490).

5. Stakeholders: Any operation involves stakeholders. The typical stakeholders for commercial operations are: customers, suppliers, employees, distributors, and the focal company. In addition to these stakeholders, humanitarian NGOs have local governmental authorities, other NGOs, donors, and media as stakeholders (Van Wassenhove, 2006, p. 477). This makes the work more complex and difficult for the humanitarian aid sector.

6. Uncertainty: Humanitarian aid, due to their dangerous and unpredictable working environment, faces much more uncertainty than the business sector. Companies usually try to minimize risk exposure in their operations. Therefore,

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TŲ UB

- Lean Applicability Assessment -

The approved original version they look for stable currencies, reliable suppliers, and stable market conditions. In mater thesis is available at the main library of the Vienna University of Technoloc contrast, humanitarian NGOs naturally face demand, supply, and assessment http://www.ub.tuwien.ac.at/eng uncertainty often at a very high level, and issues of safety are great because humanitarian NGOs may be working in a politically volatile environment (Van Wassenhove, 2006, p. 477). In addition to causing significant and frequent changes to project plans, this high uncertainty and the time pressure in emergencies lead to high staff turnover or burn-out caused by emotional and physical demands on employees. This results in a short supply of skilled staff (Van Wassenhove, 2006, p. 478).

7. Funds: In commercial organizations, funds can be raised through different channels, but typically through shareholders, bank loans, and operational profit. These funds can stay in businesses for decades. In contrast, humanitarian organization can only raise funds through donations. And there is no operational profit in a humanitarian organization, because savings (unspent funds from institutional donors) are generally frowned upon (because they are generally caused by work not being done) and must be spent or returned to the donor (Parris, Thoughts on Lean in Humanitarian Aid, 2017). Donors want to see their donations spent soon and not to be kept in the organization for long (Parris, Thoughts on Lean in Humanitarian Aid, 2017). The level of flexibility in spending money in business sector is much higher compared to the humanitarian aid sector where funds are usually earmarked for one project, and within the project for specific budget line items. This fixed budget makes it difficult to react to uncertainties for humanitarian NGOs. The only area where underspending budget is rewarded in the humanitarian sector, is overhead expense that is covered by undesignated private funding (Parris, Thoughts on Lean in Humanitarian Aid, 2017). Donors want to see their money spent in direct project expenses that create an impact. If money is not spent according to the agreed-upon budget, this is seen as negative, and the fault of the humanitarian NGO. Institutional donors generally do not cover overspending. Therefore, NGOs strive hard to spend exactly according to budget. (Parris, Thoughts on Lean in Humanitarian Aid, 2017). On the other hand, in

- Lean Applicability Assessment -

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

The approved original version business, investors reward companies for spending less than planned when it master thesis is available at the main library of the Vienna University of Technologresults in higher profits.

8. Overhead Costs: In the humanitarian aid sector, donors are interested in the overhead costs of humanitarian organizations and see a low overhead cost as an indicator of operational efficiency (Parris, Thoughts on Lean in Humanitarian Aid, 2017). Institutional donors, buyers (see Figure 3-3), only pay for a certain percentage of overhead cost, while in the commercial sector the customer does not care about the percentage of overhead cost of a product or service (Parris, Thoughts on Lean in Humanitarian Aid, 2017).

This leads to a race to the bottom of overhead costs which, in the end, reduces the efficiency of humanitarian work. Investments in information technology, systems or human capacity building are overhead expenses, which are drastically limited by institutional donors. This puts the NGO sector behind in terms of innovation. For instance, Medicines Sans Frontier (MSF) presents its overhead expenses with 6.3%, compared to the commercial sector where overhead expense is usually between 15-25% (Peterkeni & Bandara, 2015, p. 9). As another example, Charity Navigator, an organization that rates financial efficiency of NGOs, only gives the highest score only to humanitarian NGOs that spend over 92% of their budget on direct programme expense (Charity Navigator, 2017).

9. Location: For-profits can always decide where they want to operate, have their headquarters, or locate their factories and bases of operation. And they can assess and pursue the best places for expansion. In contrast, humanitarian NGOs do not have this freedom of choice. International humanitarian NGOs go to the worst crises. They are often spread throughout many different countries, cultures and continents. Nothing obliges NGOs to respond to a particular disaster, although donors, media, and the public can create pressure on NGOs. In these locations, the human resources, network of suppliers and the infrastructure (IT, roads, and other aspects) are the most limited. And these countries generally have high levels of corruption. NGOs are generally needed most – and thus operate – in the most

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

- Lean Applicability Assessment -

The approved original version of the main library of the Wienna University of Technolog to operate. (Parris, Thoughts on Lean in Humanitarian Aid, 2017)

10. Disruptiveness versus continuity: After a disaster strikes, NGOs need to react very quickly, and time is key (Eriksson, et al., 2009, p. 671). NGOs need to deploy people to identify the needs as quickly as possible, to be able to respond properly. After this happens, the programme starts to run. The workload for field programmes starts at a very high level then rises and decreases after a while, except for protracted crises where there is no common pattern of workload over time. Humanitarian programmes run until needed relief has been provided and people have recovered to their (generally fragile) condition prior to the crisis – until communities do not need the assistance of humanitarian NGOs any more. On the commercial side, businesses first start small and then try to grow continuously by developing new products or services and expanding to new markets. The contrast in development of workload over time leads to further differences which will be discussed in the assessment, in Section 3.3.3.

3.3 Assessment of Contextual Differences in Terms of Lean Applicability

3.3.1 Literature Review

To answer the first research question, if lean can benefit a humanitarian NGO, it first necessary to first assess if the humanitarian aid sector fulfils the prerequisites to be able to apply lean. There only has been made a few attempts to study lean implementation in NGOs and they are all very general and do not take the form of academic research (Pieńkowski, 2016, p. 109). Further, Pieńkowski (2016) identifies a lack of studies describing the applicability of lean in the NGO environment. Most of the existing studies only focus on the implementation. Lean4NGO.org is a good source to find studies of lean implementation in NGOs. Pieńkowski (2016, p.127) studied a lean thinking implementation of the Wrocław Food Bank, a Polish local non-profit, and discovered some impediments NGOs face in their lean application that are the similar to the humanitarian NGO context:

- Lean Applicability Assessment -

http://www.ub.tuwien.ac.at

The approved original version of this diplometer to receive donations, NGOs have to agree to terms of their donors, master thesis is available at the main library of the Vienna University of Technolog Which restricts their operational freedom to make independent choices and adjust http://www.ub.tuwien.ac.at/eng their operational models.

• Funding does not always correlate with the performance of the NGO and its organizational efficiency. Due to this, some NGOs do not realize their poor level of performance and need for improvement.

• Most NGOs do work on very limited budget and struggle to perform their core activities. This makes it difficult for them to stop fire-fighting and emphasize on continuous improvement.

Pieńkowski (2016, p.127f) concludes that due to these challenges a full lean implementation may be very difficult for NGOs, and organizations may decide to only utilize some lean tools such a visual management, standardization, and 5S. Further, Pieńkowski identifies strong leadership support and establishing a continuous improvement and problem solving culture as the key aspect of a successful lean implementation. This combination should ensure a successful lean transformation regardless of all these obstacles (Pieńkowski, 2016, p. 127).

There are similarities between local non-profit and international humanitarian NGOs, but given the different operating areas of both contexts, the humanitarian aid sector may face different challenges in lean application. Parris (2013) and GRIPS Development Forum (2009) studied how lean can be applied and adopted to the humanitarian context, but no paper has yet been published that quantified the applicability of lean for humanitarian NGOs. This paper makes an attempt to fill this gap by evaluating all ten major differences between business and humanitarian aid organizations, identified in Section 3.2, in terms of how they positively, negatively or do not influence the applicability of lean for humanitarian NGOs.

3.3.2 Methodology of Assessment

This assessment draws on the findings of existing literature and the expertise of Andrew Parris who has over ten years of experience in lean and humanitarian NGOs. These sources were used to identify the differences in context in Section 3.2. The next section

UB Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

TU UB

- Lean Applicability Assessment -

The approved originates how these differences affect on the applicability of lean in humanitarian these differences affect on the applicability of lean in humanitarian order to quantify this assessment, a 1 to 5-scale is introduced which is explained in Table 3-2. The lowest possible score, 1, means that this difference to the business world influences the applicability of lean negatively and so strongly that it limits the extent to lean can be applied and impedes humanitarian NGOs to implement lean. Score 2, influences the applicability negatively, or limits the extent of lean, but not critically. Score 3 means that the difference does not influence the applicability of lean. Score 4, enhances the lean applicability compared to the business world, and Score 5 enhances the applicability of lean strongly.

Score	Influence
1	Negative, critical
2	Negative
3	No influence
4	Positive
5	Strongly positive

Table 3-2: Score System

3.3.3 Assessment

Table 3-3 consists of three columns: the first column is the difference between commercial and humanitarian NGO sector, which is explained in Section 3.2, the second column contains the Score, and the last gives the explanation of the score.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Tech-nischen Universität Wien aufgestellt und zugänglich.

TU UB WIEN Universität

http://www.ub.tuwien.ac.at

- Lean Applicability Assessment -

Table 3-3: Lean Applicability Assessment

The approved original version of this diploma or master thesis is available at the main library of the		or	Table 5-3. Lean Applicability Assessment		
Vienna University http://www.ub.tuv	of Tech vien.ac	Different Context	Score	Explanation of Score	
1. P		Purpose	3	The different purposes are not identified to influence the applicability of lean. As long as the organization's employees strive for excellence and a culture of continuous improvement is established, it does not matter if the organization makes a profit or not.	
	2.	Staff Availability & Turnover	1	Staff turnover is identified as the most critical factor for a lean application in a humanitarian organization. As mentioned in Section 3.2, the staff turnover rate is high due to many reasons. Lean, especially the lean culture of continuous improvement, takes time and can only thrive if the employees are well trained and improve their problem solving skills through experience on the job. Toyota has realized the importance of maintaining employees and the staff turnover rate is one of their biggest concerns. The staff turnover rate at Toyota Japan is under 2% (Womack, Jones, & Roos, The Machine That Changed the World, 1990, p. 213), compared to humanitarian NGOs, for instance MSF where the rate of employees leaving after the first mission is 58.2 per cent (Korff V. P., 2012, p. 94). A high staff turnover rate leads to staff, trained in process improvement and lean, leaving the NGO and the loss of human capital, or employees leave the organization before they could be trained and establish lean thinking. Therefore, establishing and sustaining a lean culture and practice will always be challenging for humanitarian NGOs. Staff turnover is the most critical impediment to establishment a lean culture in humanitarian NGOs.	
	3.	Motivation	4	As described in Section 3.2 humanitarian aid workers are, in general, more intrinsically motivated than employees in commercial organizations. This means that if an aid worker can see the benefits of lean and internalizes the concept, great results and improvements can be achieved because lean thrives through intrinsic motivation. Lean thinking can benefit from this higher	

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbiblioth - Lean Applicability Assessment -

The approved ori	hinal version of this diploma	lor	
master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng		the	significantly enhances the development of a lean culture.
	4. Customer	2	Value identification is critical to lean and is done by defining process steps as value adding or waste from the customer perspective (Pieńkowski, 2016, p. 114). As shown in Figure 3-3, the benefitting customer (the beneficiary) does not pay for the service in the humanitarian sector. To ensure the future of the organization, NGOs have to satisfy donors and not only beneficiaries. This means that a single customer focus is not enough for humanitarian NGOs. The context is more complex compared to the commercial sector, which makes it more difficult to apply lean and is an obstacle NGOs need to overcome.
	5. Stakeholde rs	2	A humanitarian NGO needs to consider more stakeholders in their decision making processes and operations. This has an influence on their own processes and organizational culture. Similarly, humanitarian NGOs usually cannot freely choose their suppliers or establish a sustainable relationship which limits their ability to create lean supply chains (Parris, Thoughts on Lean in Humanitarian Aid, 2017).
	6. Uncertainty	2	The commercial sector tries to avoid uncertainty and risk as much as possible. To establish a continuous flow and just-in-time delivery, a company has to have perfectly takted processes and has to collaborate with reliable suppliers. Usually, companies overcome uncertainties in supply and demand by creating inventory of raw materials and finished products (Chen, 2011, p. 874ff). According to Ohno (1998), inventory is the sixth type of waste. Humanitarian NGOs naturally operate in risky and uncertain environments. Since time is key in emergencies (Eriksson, et al., 2009, p. 671), NGOs need to be able to always deliver goods as fast as possible. This is accomplished in part through pre-positioning relief supplies at strategic locations. The high level of uncertainty humanitarian NGOs face in supply and demand, and the logistical challenges of storing and moving

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbibliothek - Lean Applicability Assessment -

The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuvien.ac.at/eng			goods in disaster contexts, do not permit just-in-time delivery or a continuous flow of goods. This limits the applicability of lean.	
7. Fu	nds	2	The incentive to save money in operations is not as appreciated or rewarded in humanitarian NGOs as compared to the business sector (Parris, Thoughts on Lean in Humanitarian Aid, 2017). Once a donor provides funds to an NGO, donors want to see their money spent and not saved. This means that improving efficiency is not really rewarded (Pieńkowski, 2016, p. 127) in the humanitarian sector. This circumstance can discourage humanitarian NGOs from making an effort to establish a lean culture.	
8. Ov Co	verhead	2	The demanded limitation on overhead costs by institutional donors also limits the potential for staff to build further skills in workshops or training, because these expenses are accounted as overhead costs. If employees are not well trained in problem solving and continuous improvement skills, a lean culture can never develop. The limitations on overhead costs cause less personnel for management and support function tasks. The workload of these employees is so high that they have little time to work on improvements. But it is just these overhead employees that define and improve processes. This hinders a culture of continuous improvement from being established (Parris, Thoughts on Lean in Humanitarian Aid, 2017). It is difficult for any humanitarian NGO to add costs on the overhead side and invest in a lean initiative, although it may provide the key to reduce overhead costs in long term.	
9. Lo	cation	2	The location of any operation, for-profit or not, limits the extent to which the supply chain can be lean, but also influences the lean organizational culture. Wangwacharakul, et al. (2014) studied the influence of the domestic culture in Sweden, and Wong (2007) in Taiwan respectively, on the lean. Both agree that culture has an impact on how lean is received and lived. Lean has to be adapted to the culture, to be received best by the employees (Wangwacharakul, Berglund, Harlin, & Gullander, 2014, p. 135).	

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Tech- nischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at		n-/ ch- th.	- Lean Applicability Assessment -
The approved originater thesis is a Vienna University of http://www.ub.tuv	ginal version of this diploma ailable at the main library of t of Technology. vien.ac.at/eng	or	In general, humanitarian NGOs consist of aid workers from different countries and cultures and local staff. The high cultural diversity within one organization makes it difficult to establish a fit for lean to be received by every employee best in the organization and be understood in the same way.
	10. Disruptiven ess versus Continuity	2	Looking at Toyota, a lean best practice company, their working environment looks very different from that of a humanitarian NGO. In Toyota City, Japan, Toyota has been producing cars since 1938 at the same location and plant (Toyota Global , 2017), and car factories generally are used for decades. This makes it easier to work on sustainable continuous improvements compared to humanitarian sector where projects usually last one to three years at one location. As shown, in Sub-item 2, Staff Turnover, only 40% of international staff reenlists for a next project. In the disruptive way, humanitarian operations run, designed as project work, it is challenging to establish sustainable process improvement, and receive benefits from it, if processes are needed for a short period of time.

3.3.4 Discussion

TU UB

The arithmetic mean score of the ten identified differences is 2.2. The low value leads to the conclusion that the differences of humanitarian NGOs to the business organizations rather impede than enhance lean in this sector. The different purposes of business and humanitarian aid (1) are not identified to influence the applicability of lean.

Staff turnover and availability, with the lowest possible score 1, is identified as the most critical impediment to a full lean implementation. The problem of employee turnover for lean application and implementation is that establishing an organizational culture which is based on training and skills, is very challenging if employees leave shortly after being trained or even before receiving the necessary training. This aligns with the suggestion of Stamm & Neitzert (2008, p.9) who suggest employee turnover rate as Key Performance

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TU UB

- Lean Applicability Assessment -

The approved original version (KPI), for a lean implementation. Womack et al. (1990) also identified staff with the main library of the most critical KPI for Toyota.

Only one out of the ten differences, the higher tendency for intrinsic motivation for employees in humanitarian NGOs, is identified to enhance lean in this sector. To be able to profit from this potential benefit, a humanitarian NGO has to train people extensively and establish lean thinkers.

The different customer-buyer relationship (4, see Figure 3-3), the lower incentive to improve efficiency (7), the limit on resources for staff development (8), the far geographical stretch and high cultural diversity in humanitarian NGOs (9), and the highly disruptive work environment (10) are identified as obstacles which make it more difficult to implement lean for a humanitarian organization. The financial pressure prevents humanitarian organizations to invest in general, but also in staff capability building. This combined with donors' low appreciation of improvements in efficiency, create one the hand a huge impediments and on the other, a drastically low incentive for humanitarian NGOs to become a lean organization. This circumstance may explain why lean has not been established in this sector.

Dealing with a higher number of stakeholders and not being able to freely choose their working partners (5), and the high level of uncertainty humanitarian NGOs face (7), are identified as limiting factors for a full lean application. For example, lean tools may improve the supply chain efficiency for humanitarian NGOs, but due to time limitations, the low number of possible suppliers, and high uncertainty, the supply chain will never become as lean as Toyota's.

3.3.5 Conclusion

To conclude, humanitarian NGOs face significant challenges, obstacles and limitations in becoming lean that do not exist for the business sector. The differences to the business sector overall make it much harder to implement lean in the humanitarian sector. Nevertheless, it would be wrong to conclude that because of the existing impediments and limitations lean cannot benefit humanitarian NGOs. This conclusion only states that

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbiblic

- Lean Applicability Assessment -

The approved original vision original vision or challenging for humanitarian NGOs to become lean and lean cannot be viena university applied to the same extent as for commercial companies like Toyota.

The approved original vedor Potential Benefit and Need of Implementing Lean for Vienna University of Technology. http://www.ub.tuwien.ac.tieng Humanitarian NGOs

In order to answer the research question, if lean can benefit a humanitarian NGO, Chapter 1 assesses the ease of application of lean for humanitarian NGOs. Considering the different context, there are challenges to apply lean in humanitarian NGOs, identified in Section 3.3.3, and lean can only be applied to a limited extent. Still, the above chapter does not include the potential benefit of lean for NGOs and the need for NGOs to apply lean.

Let us consider that a humanitarian NGO can overcome the challenges to start a lean initiative, and now asks how can lean be adopted and implemented to be most beneficial for their operational performance. Lean offers various improvement tools and an organizational culture of continuous improvement (see Section 2.2.2). The first part of this chapter contains a lean tool assessment for humanitarian aid purposes. Section 4.1.1 contains a literature review of lean tools that have already been implemented in the humanitarian aid sector. Section 4.2 investigates if the lean culture of continuous improvement can benefit humanitarian agencies. Section 4.3 identifies that there is a need for lean in the humanitarian aid sector.

4.1 Lean Tools for NGOs

Work content in humanitarian aid usually occurs in two very different parts. One part is office work. Donors demand high transparency and therefore humanitarian NGOs must document and account all orders and expenses carefully. Marketing, writing proposals, human resource, logistics, and finance work occur in offices. The other part, operations, is very different. Staff members go out to the field to build homes, dig drinking water wells, provide medical care, give training, or hand out aid kits, and nutrition. As discussed in Section 2.2.2, lean offers tools to improve efficiency and performance of operations and office work.

Chandra (2013) summarizes the twelve most critical lean manufacturing tools to know which are shown in Table 4-1. One lean tool, value stream analysis, is added to these 12

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved origination of the view of th

applications in literature, given in Section 4.1.1, and the expertise of Andrew Parris.

Lean Tool	Description	Applicable for
		humanitarian NGO:
1. Cellular Manufacturing	Cellular manufacturing is an approach in which all equipment and workstations are arranged based on a group of different processes located in close proximity to manufacture a group of similar products. The primary purpose of cellular manufacturing is to reduce cycle time and inventories to meet market response times.	Partly. Humanitarian NGOs do not manufacture, but distributions of goods can resemble an assembly plant.
2. Takt Time	This is the "heartbeat" of the customer. Takt time is the average rate at which a company must produce a product or execute transactions based on the customer's requirements and available working time.	Yes. A useful methodology to investigate capacity and find bottle necks for processes in humanitarian NGOs.
3. Standardized Work	A documented description of methods, materials, tools, and processing times required to meet takt time for any given job. This aids in standardizing the tasks throughout the value stream.	Yes. Very useful to eliminate variance and mistakes in processes for NGOs. It makes work simpler and easier to learn, which is beneficial for high turnover rates.
4. One Piece Flow or Continuous Flow	This concept emphasizes reducing the batch size in order to eliminate system constraints. A methodology by which a product or information is produced by moving at a consistent pace from one	Partly. Small batch sizes and continuous flow improve operational performance

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

TU UB WIEN Universitätsbibliothek

he approved	original version of this diploma or	value-added processing step to the pext	for NGOs However
ttp://www.ub	s available at the main library of the ity of Technology. .tuwien.ac.at/eng	with no delays in between.	procurement and transportation over long
			distances and distributions can make small batches inefficient.
	5. Pull Systems and Kanban	A methodology by which a customer process signals a supplying process to produce a product or information or deliver product/information when it is needed. Kanban is the signal used within a pull system through scheduling combined with travelling instruction by simple visual devices like cards or containers.	Partly. Beneficiaries needs are assessed and then a humanitarian NGO can deliver supply, but with a long lead time. Still, preposition is necessary. Very uneven demand limits applicability.
	6. Five Why's	A thought process by which the question "why" is asked repeatedly to get to the root cause of a problem.	Yes. Methodology applicable to every sector and problem to find the root cause of problems in any process.
	7. Quick Changeover / SMED	A 3-stage methodology developed by Shigeo Shingo that reduces the time to changeover a machine by externalizing and streamlining steps. Shorter changeover times are used to reduce batch sizes and produce just-in-time. This concept aids in reducing the setup time to improve flexibility and responsiveness to customer changes.	Partly. Taking this concept out of manufacturing, streamlining and smaller batches are beneficial to a humanitarian NGOs as well.
	8. Mistake Proofing / Poka Yoke	A methodology that prevents an operator from making an error by incorporating preventive in-built responsiveness within the design of product or production process.	Yes. Applicable to many processes including logistics, and filling in forms.
	9. Heijunka / Leveling the Workload	The idea that, although customer order patterns may be quite variable, all of our processes should build consistent	Partly.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

TU	UB
WIEN	Universitätsbibliothek

he approved aster thesis i ienna Univers ttp://www.ub	original version of this diploma or s available at the main library of the ity of Technology. .tuwien.ac.at/eng	quantities of work over time (day to day, hour to hour). This strategy is adopted by intelligently planning different product mix, and its volumes over period of times.	Levelling workload is often not possible with the unpredictable nature of humanitarian needs.
	10.Total Productive Maintenance (TPM)	A team-based system for improving Overall Equipment Effectiveness (OEE), which includes availability, performance, and quality. This aids in establishing a strategy for creating employee ownership autonomously for maintenance of equipment. The goal of the TPM program is to markedly increase production while at the same time increasing employee morale and job satisfaction.	Partly. This can mainly be applied in fleet management (internal) and with wells and other equipment that is transferred to communities to use (all of which require maintenance).
	11.Five S	5S is a five step methodology aimed at creating and maintaining an organized visual workplace. This system aids in organizing, cleaning, developing, and sustaining a productive work environment. 5S stands for sort, set in order, shine, standardize, and sustain.	Yes. Methodology is applicable for every work environment, especially offices and storages.
	12.Problem Solving / Plan- Do-Check-Act (PDCA)	The PDCA cycle is four step method for solving problems. It helps to think that every activity and job is part of a process, that each stage has a customer and that the improvement cycle will send a superior product or service to the final customer.	Yes. A general methodology to solve problems and improve processes that is not specific to any sector, but applicable to every work process.
	13.Value Stream Analysis Process Mapping	VSA focuses on the analysis of specific value streams. It breaks one process down into many process steps which can be analyzed separately in a manageable manner.	Yes. A very powerful tool to analyze and improve any process regardless the context.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Tech-nischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

TUUB

4.1.1 of this d Literature Review of Existing Implementations

http://www.ub.tuvThisatsection provides a review of lean tools and concepts that have already been implemented in the humanitarian aid sector.

4.1.1.1 Lean Six Sigma in World Vision East Africa (Parris, Fighting Poverty in East Africa with Lean Six Sigma, 2015)

WVI was the first development and humanitarian NGO to ever start a major lean six sigma initiative in their East Africa region, as mentioned in the introduction. A lean six sigma master black belt was hired to lead the lean initiative and train lean six sigma green and black belts. An explanation of the green and black belt training is given in Section 6.2.2. The work done was lean six sigma, but primarily lean. The initiative at WVI shows that lean can have great success in NGOs. The initiative resulted in 56 trained LSS green and 11 LSS black belts. Honsha, a lean consulting group, helped to find volunteer trainers for some of the black belt training which shows that the commercial lean practitioners have interest in providing their experience for free. The benefits of this lean six sigma initiative for WVI are:

- 1.5 Mio USD reduction of annual costs
- An average of 59% reduction of span time of targeted processes
- Reduced of underspending, better reputation, mobilization of resources, and transformed culture

This example proves that lean six sigma can benefit a development NGO. WVI stopped the initiative because of a lack of leadership buy-in.

Another example of lean in East Africa is the lean implementation in the Comprehensive Community Based Rehabilitation (CCBRT) in Tanzania, a governmental health institution. Michael Grogan, a lean six sigma expert, worked at CCBRT for two years and lean showed success at their clinic. For further information. see: https://www.lean.org/LeanPost/Posting.cfm?LeanPostId=342 (2017).

4.1.1.2 Food for the Hungry in South Sudan (Ndihokubwayo, 2015)

Darine Ndihokubwayo, a lean six sigma black belt candidate trained in WVI, showed in her project that lean tools can be applied to benefit food distributions. In her project she
 Image: UB

 Die approbierte Originalversion dieser Diplom-/

 Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

UB

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved original sector visual management to make it easier visible for beneficiaries where they have to mater thesis is available at the main library of the Vienna University queue depending on their household size. She mapped the current distribution process, http://www.ub.tuwien.ac.at/eng found waste and improved it, made a new distribution area layout and rearranged batch

sizes in order to simpler distribute the goods in the right amount. These changes resulted in:

- Fewer process steps for beneficiaries to get food
- Reduced waiting time
- Reduced handling costs
- Reduced workload for aid workers
- Less effort for reporting and control

This application of the lean tool, process mapping, worked for food distribution in humanitarian aid.

4.1.1.3 Toyota Meals per Hour (EL-Naggar, 2013)

The Food Bank for New York is the biggest anti-hunger charity in the United States of America, feeding over 1.5 million people a year. The Food Bank is mostly funded by businesses, but Toyota had a new idea for helping Food Bank. Instead of donating money, Toyota showed them how to improve their processes. Toyota engineers applied lean tools to streamline their process. The achieved great results:

- Reduced dinner waiting time from 90 to 18 minutes in their Harlem soup kitchen
- Reduced time to fill bags from 11 to 6 minutes in a Staten Island food pantry
- Reduced time to pack a box of supplies for victims of the Hurricane Sandy from 180 to 11 seconds in a warehouse in Bushwick, Brooklyn.

This is an example another example of how lean helped in distribution of non-profits.

4.1.1.4 Value Stream/Chain Mapping (Taylor, 2009, p. 27ff)

The theoretical framework on which this application of lean is built is called 'lean thinking'. Lean offers methodologies to improve processes in procurement, distribution, supply chain management, and manufacturing. Lean has spread through different sectors http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved original begins b

Value chain analysis was applied for a humanitarian was applied to improve the supply chain of an emergency kit with twelve items for Africa. KPIs of the supply chain were identified, then the current state was mapped, after that four different alternative supply chains, and an improved version of the existing supply chain were mapped. The conclusion states that "Lean and value chain analysis have significant potential to facilitate the improvement of existing international humanitarian supply chains and in helping to develop efficient African-based alternative supply systems. (Taylor, 2009, p. 29)"

Another implementation of a value chain analysis of an Indian charity in 2014 can be found at lean4ngo.org.

4.1.1.5 Lean Humanitarian Logistics

Cozzolino et al. (2012) did a case study on the United Nation Worlds Food Programme to examine if agile and lean principles are needed in the humanitarian aid supply chain and at what specific stages. Lean and agile principles are both needed in specific stages of the humanitarian aid logistics process (Cozzolino, Rossi, & Conforti, 2012, p. 29). Lean principles are needed in the reconstruction phase because there the objective is efficiency, while agile principles are needed in the stage right after the disaster (Cozzolino, Rossi, & Conforti, 2012, p. 28). Still, there has not yet been a case study on a lean supply chain in humanitarian aid. Probably because of the reasons discussed in Section 3.3.3.

4.1.1.6 Five S (Grunden, 2013)

Global Links is a medical relief and development NGO. The organization applied 5S to a storage area of medical equipment, such as suction machines, ophthalmoscopes, nebulizers, surgical clippers, electric thermometers, and blood pressure equipment. The room was not organized and it took employees and volunteers a long time to find equipment which wasted time and caused frustration of staff. The 5S method was used to resolve this problem.

 Image: UB

 Die approbierte Originalversion dieser Diplom-/

 Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

HUUB

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved original version of this disport or master thesis is available at the main library of the Vienna University of Technolog posted to right places, since not all staff knew how a nebulizer looks like.

2. Set in order: The equipment was eyeballed. Some of the equipment was not working, has run out of date, or was not useful and then sorted out.

3. Shine: The storage room was cleaned and equipment was put into place according to their pictures. One shelf was identified for incoming items, and staff trained how to identify, clean and put those items away.

4. Standardize: The handling and placement of items was standardized and the color-coding and photographs helped to avoid mistakes.

5. Sustain: This was more interpreted as continuously improving by identifying large equipment that was used rarely and took a lot of the limited storage space. Such items were replaced by smaller items with constant demand.

Garcia, the deputy director of Global Links, concluded: "Staff time and volunteer time are precious commodities for an organization like Global Links. Finding new and creative ways to reduce wasted time not only improves their efficiency, but shows respect to the people who work there, and ultimately, to their recipients." Further she concluded that 5S even helped Global Links to focus better on their mission by providing more frequently demanded items than before.

4.1.1.7 Interest in Lean for Humanitarian Work

There are several institutions which are offering lean training, especially for humanitarian NGOs and are interested in bringing lean or six sigma to humanitarian aid. The following lists the three most important ones and their websites to look for further information:

- Lean4NGO: http://lean4ngo.org
- NGO pulse: http://www.ngopulse.org
- Lean Six Sigma for Good: http://www.leansixsigmaforgood.com

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

TUUB WIEN Universitätsb

4.2 Lean Thinking for Humanitarian NGOs

master thesis is available at the main Vienna University of Technology.

http://www.ub.tu^TThe answer to the question if lean thinking is applicable to humanitarian aid context is given by James P. Womack, former President of the Lean Enterprise Institute who states: "Lean thinking works everywhere!" (2006). This work draws on the idea that an organizational culture that has continuous improvement of processes and quality as main daily tasks, is not applicable specifically to manufacturing or service industry, but to all organizations in any field. Mike Rother describes such a lean thinking organization in the introduction of his book, *Toyota Kata*: "Imagine you have a way of managing that generates initiative among everyone in the organization to adapt, improve, and keep the organization moving forward." (2009, p. 3). This quote does not give information on the specific work field. It describes a mindset of an organization. Section 4.2 investigates how

4.2.1 Implementing Lean Thinking

to best implement lean in the humanitarian aid sector.

Organizations that want to implement lean thinking must undergo a process of significant organizational change (Helten & Lindemann, 2012, p. 51). Two main types of models exist for organizational change: prescriptive and emergent change (Liu, 2009, p. 234). The prescriptive change is planned, while the emergent change takes place unintendedly, due to ongoing adaptations (Liu, 2009, p. 51). Lean change is prescriptive. Liker (2003) describes in his book, *The Toyota Way*, Toyota's business principles which are shown in Figure 4-1.



Figure 4-1: Business Principles of Toyota (Liker J., The Toyota Way: 14 Management Principles from the World's Greatest Manufacturer, 2003)

HUUB

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved original Toyota, builds, cap acity in people and business partners to enable the organization and mater thesis is available at the main library of the chain to establish a culture of continuous improvement. If an organization applies lean tools and all employees are lean thinkers, it has successfully established a lean culture (Parris, Thoughts on Lean in Humanitarian Aid, 2017). Zarbo (2012) quotes Liker (2009) that 90% of those who attempt lean, fail. The explanation for this is that many organizations fail to understand Toyota's culture which is dedicated to developing human talent throughout the enterprise to continuously improve the organization (Zarbo, 2012, p. 321). This means that implementing lean tools is easier than establishing lean thinking. But why?

To become a lean, an organization has to understand that continuous improvement is a daily task. As Akers says: "Lean is hard work that makes everything else easy. (2016, p.96)" One major obstacle is the duration of an organizational culture change such as lean. Establishing a full lean culture takes about a decade (Zarbo, 2012, p. 321). This long duration of change may cause frustration of team members and executives (Zarbo, 2012, p. 321). It is critical to be patient, keep energy behind the initiative the whole time, and further develop lean in the organization. One problem can be that during a lean implementation the executive leader changes and the new Chief Executive Officer (CEO) wants to implement a different organizational culture which may destroy all the efforts that have been achieved during the lean implementation.

Lean must be incorporated into the existing business model to further support the organization's strategic objectives. Every implementation will be unique due to existing culture, organizational values, context, organizational structure and other factors. There exist no standard step-by-step guidelines, no textbooks to follow. (Ortiz, 2012, p. 151)

4.2.2 Lean Leadership

The missing link between waste reduction through lean tools and a continuously improving lean organization could be leadership (Dombrowski & Mielke, 2013, p. 565). The most critical factor for a lean successful lean thinking implementation (Byrne, 2017, Pieńkowski, 2016 & Zarbo, 2012).

UB

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

Although improvements are made by every employee, leadership has to motivate and master thesis is available at the main library of the Vienna University encourage employees to work on improvements (Zarbo, 2012, p. 323). Mike Rother's, http://www.ub.tuwien.ac.at/eng *Toyota Kata*, studies Toyota's culture and how leaders act as mentors. At Toyota, lean leaders spend more than 50% of their time on coaching people. It is the main part of their supervisor's evaluation (Rother, 2009, p. 188f). Coaching is something organizations have to realize and implement thoroughly when they apply lean.

Lean leadership has 5 major characteristics (Balle, 2009):

1. Challenge employees: Giving employees difficult tasks, shows that a leader respects them and trusts their abilities to find good solutions

2. Kaizen: Good enough does not exist. No process is ever perfect and there is always room for improvement

- 3. Go and See: Going to the source of the problem
- 4. Develop: Coach employees to develop their skills

5. Teamwork: Clear responsibilities for individuals and teams to maximize performance and build mutual trust

To show the difference between a lean and a conventional leader, Parris (2017) defines eight different leadership traits or habits between a conventional and a lean/continuous improvement (CI) leader, shown in Table 4-2.

Conventional Leader	Lean Leader
Delegate Lean/CI	Personally lead Lean/Cl
Lean/CI is secondary	Lean/CI is primary
From a distance	From the Gemba
Problems are bad	"No problem" is bad
Solve problems	Grow problem solvers
Give answers	Ask questions
Fight fires	Find and address root cause
Result oriented	Process and results oriented

Table 4-2: Different Leadership Traits (Parris, Thoughts on Lean in Humanitarian Aid, 2017)

HUUB

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved original tendences are the sis available at the main library of the vienna University traits during a lean implementation. Some leaders may not be in favor of lean from the had bad experiences or heard stories of failure. Therefore, it is very important to convince and incentivize all leaders to get on board a lean implementation, in order to be successful. Leaders have to be the ones pushing the initiative (Zarbo, 2012, p. 323).

4.3 Need of Lean for Humanitarian NGOs

This section contains the need of lean in the humanitarian aid. There are incentives for humanitarian NGOs to become lean:

• Lean has enabled many organizations to increase their performance. Humanitarian NGOs that have better processes in place are able to increase their performance, which leads to the ability to fulfil their mission better and reach more beneficiaries. Due to the context humanitarian NGOs work in, the application of lean does not only impact NGO's efficiency, service quality or effectiveness, but lean can enable humanitarian NGOs to save more lives and relieve more humans from suffering around the world. Lean, implemented and adapted in the right way, can improve the standard of living for the poorest in the world.

• As mentioned in the introduction, the word lean comes from doing more with less. Efficiency is very important for humanitarian NGOs, especially because of the common donor opinion that overhead costs are a sign of efficiency. In order to reduce overhead, processes need to be improved and made easier, so that less time is needed to fulfill the same tasks. As shown in Section 4.1, lean provides many process improvement tools which can help to reduce time of processes, and as a result reduce overhead costs. Low overhead costs attract donors which leads to growth for humanitarian NGOs.

• Lean helps to simplify processes, which is very helpful considering the high turnover rate humanitarian NGOs have. If processes are simple, new employees can be introduced faster into their new tasks. Simple processes ease the work of employees and make them feel better about their work which results in less http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

TUUB

The approved original version and higher job satisfaction (Parris, Thoughts on Lean in Humanitarian The approved original version of the Vienna University of Technolog Aid, 2017). Legnerova (2016) did a study on demotivation factors for non-profit http://www.ub.tuwien.ac.at/eng organizations and the results show that 'difficulty of work' (62%) is the most critical factor for demotivation under non-profit staff, even higher than 'low financial reward' (51%), and 'manager`s behavior' (20%). To ease the work for employees is one of the most important benefits of lean for humanitarian NGOs that can be used to address the high staff turnover rate.

4.3.1 Lean and the Core Humanitarian Standard

Quality work in humanitarian aid which will now be certified by an institution called the Core Humanitarian Standard (CHS). The CHS consists of nine different commitments and quality criteria (Core Humanitarian Standard, 2014, p. 9), shown in Figure 4-2. "The CHS describes the basic essential elements of principled, accountable and high-quality humanitarian action (CHS Alliance, 2016)." To prevent any conflict of interest the certification for CHS is verified by the independent organization the Humanitarian Quality Assurance Initiative (HQAI, 2017). In terms of how the CHS aligns with lean, special attention should be given to criteria 7 which is about continuous improvement, the core of the lean organization, as discussed in Section 4.2.1.

TU UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Tech-nischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -TU UB



Figure 4-2: CHS Standards on Quality and Accountability (Core Humanitarian Standard, 2014, p. 4)

Parris (2017) investigated the nine commitments of CHS in terms of how these fit with the principles of lean, shown in Table 4-3. The CHS aligns well with the principles of lean (Parris, Thoughts on Lean in Humanitarian Aid, 2017). The term Process Excellence, used in the table, is an adapted version of lean for the humanitarian context by Parris. Every process of a humanitarian organization should be (Parris, Green Belt Training, 2008, p. 112):



Figure 4-3: Process Excellence

- 43 -

TUUB

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The approved original a process fulfils all attributes shown in Figure 4-3, it can be considered to be excellent. master thesis is available at the main library of the Vienna University That does not mean it has reached a state of perfection, because there is always room http://www.ub.tuwien.ac.at/eng for improvement. A more detailed explanation of process excellence can be found in the

Appendix. In Table 4-3, Parris shows how the five attributes of Process Excellence (basically lean) match the CHS principles.

Table 4-3: Process Excellence Match with CHS (Parris, Thoughts on Lean in Humanitarian Aid, 2017)

Process Excellence on CHS Commitments	Effec tive	Effici ent	Appro priate	Empo wering	Contin uously Improv
					ing
1. Communities and people affected by crisis receive assistance appropriate and relevant to their needs.	>		~		
2. Communities and people affected by crisis have access to the humanitarian assistance they need at the right time.		✓			
3. Communities and people affected by crisis are not negatively affected and are more prepared, resilient and less at-risk as a result of humanitarian action.	~			✓	
4. Communities and people affected by crisis know their rights and entitlements, have access to information and participate in decisions that affect them.		~		✓	
5. Communities and people affected by crisis have access to safe and responsive mechanisms to handle complaints.			✓		✓
6. Communities and people affected by crisis receive coordinated, complementary assistance.		✓		✓	
7. Communities and people affected by crisis can expect delivery of improved assistance as organisations learn from experience and reflection.					✓
8. Communities and people affected by crisis receive the assistance they require from competent and well-managed staff and volunteers.		✓		✓	
9. Communities and people affected by crisis can expect that the organisations assisting them are managing resources effectively, efficiently and ethically.	✓	~	✓		

http://www.ub.tuwien.ac.at - Potential Benefit and Need of Implementing Lean for Humanitarian NGOs -

The assessment of Parris (2017) shows lean or the humanitarian NGO adaption of lean, mater these is available at the main library of the Vienna University Process Excellence, aligns very closely with the CHS and thus can help get a CHS http://www.ub.tuvien.ac.at/eng certification. Especially, looking at number 7, about continuous improvement, the core of lean. Quality certifications are of high importance to NGOs, because they are recognized by donors. CHS has obviously realized the need of continuous improvement in humanitarian aid, like the International Organization for Standardization (ISO) 9001 certification for quality management, has continual improvement as requirement number 8.5.1 (ISO 9001 Checklist, 2015).

4.4 Conclusion

The lean tools assessment for humanitarian NGOs in Section 4.1 shows that many tools that were originally developed in a very different context can benefit humanitarian NGOs too. They are not all fully applicable as at Toyota, but can be adapted to the humanitarian aid context. The literature review shows that there are not many applications of lean in the humanitarian context yet, but lean already showed success in this sector. The few applications may be because of the entry barriers for lean in the humanitarian aid sector, discussed in Section 3.3.3. Implementing lean thinking and in long-term an organizational culture of continuous improvement is shown to be beneficial for humanitarian NGOs. Section 4.3 shows that there is a clear need to establish lean in the humanitarian aid sector. Donors and standards institutions for humanitarian NGOs have realized that continuous improvement and organizational learning is necessary. Implementing lean is a proven and effective way to establish this culture of continuous improvement. Humanitarian NGOs are under great pressure and have very limited resources to improve their performance, given the low funds for investments. Lean implementation is a low cost way to improve efficiency, given that no expensive investments have to be made. Lean implementation takes time and steady managerial support to be sustainable, as discussed in Section 4.2. Overall, this chapter shows that lean provides enormous potential benefits for humanitarian NGOs, but it needs to be adapted and implemented carefully. The lean implementation always needs active participation from all employees, starting with senior leaders and extending to front-line workers, to be successful.



5 Case Study: Medair

/ienna University of Technology

Medair, a humanitarian international non-governmental organization (INGO), started a lean initiative in 2017 which is studied in this section. Section 5.1 provides information about the organization's identity, operations, financials, and structure. Section 5.2 assesses the unique context and challenges the organization faces.

5.1 Overview of Organization Under Study

Medair is a Swiss non-governmental organization delivering humanitarian aid globally. Founded in 1989, out of a single project in Uganda, it developed into a globally operating organization, responding to crises in 14 different countries, delivering aid to over 1.8 million people (Medair, 2016). It is a faith-based organization, responding to humanitarian crises, especially, in remote areas.

5.1.1 Vision & Mission

Medair's vision, released in 2017: "That the world's most vulnerable and difficult to reach people live in dignity, free from human suffering and with hope for a better future." According to Collins and Porras (1996), the vision statement represents the organization's envisioned future. For Medair this means that there is no human suffering anymore and people have hope of change for the better. To achieve this goal, the company needs to work towards achieving this state. The desired state is defined by the vision.

In contrast, the mission states the core purposes of the company and therefore provides reasons for its existence. It defines a company's ideology and character and is the glue that holds it together (Collins & Porras, 1996). Medair's new mission statement (2017): "Medair is an impartial, independent, and neutral humanitarian organization inspired by Christian faith to relieve human suffering in the world's most difficult-to-reach and devastated places."

As discussed in Section 3.1.3, the principles of humanity, impartiality, independence and neutrality are derived from, the Fundamental Principles of the International Red Cross

- Case Study: Medair -

TU UB

The approved order and Red Crescent Movement proclaimed in Vienna in 1965. In addition, the adjectives matter their is available at the main thrary of the Viena University "impartial" and "neutral" resemble two of the three principles of the humanitarian space model developed by Tomasini and Van Wassenhove (2004), shown in Figure 3-2: Humanitarian Space (Van Wassenhove, 2006, p.478). The third principle, "humanity ", is mentioned in the second sentence. The humanitarian space model defines the context in which humanitarian aid is possible. This means that the mission statement characterizes Medair as a humanitarian aid organization. The organization's ideology lies in its Christian faith. Helping those in need is one of the cornerstones in Christianity. A Bible verse that shows this, can be found in Luke (10:27b): "Love your neighbor as yourself." There are other parts in the Bible that instruct Christians to care for those in need. International Medair employees have to be committed Christians which inspires Medair staff to help people in need.

One strategic decision, mentioned at the end of the mission statement: "(...) in the world's most difficult-to-reach and devastated places.", is to concentrate its efforts on the most remote and devastated places in the world. The operating conditions in humanitarian aid are especially complex (Van Wassenhove, 2006, p.477). Medair has committed itself to focus on the hardest-to-reach places in this already critical operating environment – a highly challenging task. Furthermore, strategically this means that the organization focuses on a niche in the humanitarian aid sector. The costs and efforts to reach beneficiaries are therefore higher and the environment is often even more dangerous.

5.1.2 Growth & Financials (Medair, 2016)

The operating income in 2015 amounted to 64 Million USD, whereas in 2012 it had been 39 Million USD, which shows the rapid growth of Medair in income of about 64.8% in three years. The humanitarian aid sector has been growing, for three consecutive years, in funding from 18 Billion USD in 2012 to 28 Billion USD in 2015, see Figure 5-1, which is an increase of about 55.6% in the same period. This shows that Medair has been growing faster than the overall humanitarian aid sector, but still is a small organization in the market with a market share of 0.23%.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



Figure 5-1: International Humanitarian Response, 2011-2015 (Development Initiatives, 2016, p.36)

The share between institutional donors and private donors is very important for Medair. In 2016, 18.4% of the funds Medair raised came from private donors, while the humanitarian sector average is 22.1%. Private donations are vital to humanitarian organizations, because they are usually not dedicated to specific projects. These unrestricted funds give organizations freedom to spend them where they are most needed (on underfunded budget items). On the other hand, institutional donors fund specific projects and money has to be spent only for designated "allowable" costs. This is important because institutional donors generally underfund, or sometimes do not fund, indirect or overhead costs, and unrestricted private donations are the only possible source of funds to cover these costs. Further details on funds are discussed in Section 3.2.

Overhead costs are the costs for coordinating projects, marketing, fundraising, human resources, and other efforts that are necessary to operate successfully, but do not directly help the beneficiary. In Medair, the overhead costs are divided between fundraising activities and general management, which accounted for 3% and 7%, respectively, of the operating expenses in 2016. Figure 5-2 shows the distribution of costs. Costs, which occur at the headquarters, which are humanitarian expense, but indirect for instance, shelter assistance management, accounted for 7% in 2016; 83% was spent on direct humanitarian expense meaning at the field.



- Case Study: Medair -



Figure 5-2: Operational Expense Medair 2016

5.1.3 Working Areas and Organizational Structure

Medair runs 14 different country programmes, international headquarters (HQ) in Ecublens, Switzerland, and six affiliate offices. Figure 5-3 shows a more detailed overview of the organization. The HQ with around 110 employees in Ecublens supports the country programmes in staff, marketing, recruitment, logistics, and finance. Affiliates` main tasks are raising funds, marketing, finding new donors, and attracting staff.



Figure 5-3: Medair Overview (Medair, 2016)

Country programmes have a main base, usually in the capital or a big city of the respective country, to manage all legal issues and then one or more bases in the areas they work in. Medair runs only humanitarian aid projects. Some may consider its work in

- Case Study: Medair -

TU UB WIEN Universitätsbiblio

The approved original development because of the project's length, but it helps people prepare inster thesis is available at the main library of the Vienna University for hand respond to recurring crises. The difference between humanitarian aid and http://www.ub.tuvien.ac.at/eng development work lies in the focus. While the aim of humanitarian aid is to prevent human casualties in conflicts and (slow or sudden onset) disasters and to assure the basics for survival, development work assistance aims to reconstruct infrastructure, institutions, and economy after a disaster or war (Branczik, 2004), as discussed in Section 3.1.2. Therefore, humanitarian aid projects can transform over time into development projects (and vice-versa), but there is no clear distinctive line between them.

According to the identified need in the region, Medair offers different assistance, shown in Table 5-1. The biggest operating area by beneficiaries is Health & Nutrition.

Programmes	Activities	Direct Beneficiaries
Health & Nutrition	 Treat patients at Medair supported clinics Teach people on life-saving health/nutr. practices 	989.002448.657
	Treat malnourished patients	• 35.585
Water, Sanitary & Hygiene	 Deliver safe drinking water access Deliver access to new latrine or bathing facility 	416.597122.405
	• Teach people about life-saving hygiene practices	• 276.724
Shelter & Infrastructure	 Improve and build infrastructure Teach people construction methods Deliver shelter assistance 	 229.057 16.994 13.459
Cash Assistance	Provide cash assistance for people in need	• 43.298

Table 5-1: Medair Programmes (Medair, 2016)

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

5.1.3.1

http://www.ub.tuwien.ac.at

- Case Study: Medair -

TU UB WIEN Universitätsbiblioth

Organizational Structure at Headquarters

The security of the organization and makes major decisions, for instance to deploy a team into an emergency or open a new country programme, and reports to the advisory board. The Executive Office Director has only the Process Excellence Manger who reports to him. This relationship will be explained more detailed in Section 6.1.2.



Figure 5-4: Organizational Chart Medair (July 2017)

A functional organizational structure has typical advantages: efficiency and high specialization, and disadvantages: silo-thinking and high degree of bureaucracy (Davoren, 2016). To achieve the gain from the benefits of a functional structure, it is necessary that every role in the system is understood and that staff are experienced in their job positions (Joseph, 2017). This is a problem for Medair, because staff turnover is high. It needs time for new staff to know all the necessary bureaucratic steps and to be efficient in fulfilling tasks in this system. To become a specialist, experience is critical.

- Case Study: Medair -

The approved original light turnover reduces the benefits of this organizational structure and aggravates its master thesis is available at the main library of the Vienna University disadvantages.

Another disadvantage of this organizational structure is that if the organization grows, it can be a challenge to maintain control, because decision-making has to be delegated and functional areas become more and more independent, making coordination among areas more difficult (Joseph, 2017).

5.1.3.2 Organizational Structure at Field Programmes

Around 1100 of the 1200 Medair employees work at the field. At this level Medair distinguishes between international recruited staff and national recruited staff. National recruited staff are usually members of the communities Medair works in. They do not necessarily have to have high skills or education, but play an important part in the success of country programmes. In some programmes, Afghanistan for instance, international recruited staff is not allowed to leave the compound, due to the risk of kidnapping. Therefore, national staff are sometimes the only way to distribute goods and provide services to beneficiaries, meaning that national staff execute most operational tasks in country programmes.

In contrast, international staff are recruited for special skills, and usually hold managing positions in programmes. Staff come from all over the world and are selected through various stages. Medair selects applicants according to their organizational fit. Applicants need to own the organization's values.

5.2 Unique Humanitarian Context – Medair

Knowing now what the organization does and its values, it is time to specify the unique context for Medair. Medair is an international faith-based NGO, according to Section 3.1.1, a third-sector organization demonstrates the stepwise differentiation to the unique context. Table 5-2 shows a step-wise differentiation to the specific context Medair works in.

UB Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TU UB		
Die approbierte Originalversion Masterarbeit ist in der Hauptbib nischen Universität Wien aufgestel	dieser Diplom-/ Jliothek der Tech- llt und zugänglich.	
http://www.ub.tuwien.ac.at	-	Case Study: Medair -
	Universitätsbibliothek	able 5-2. Medair Characteristics
The approved original version o	f this diploma or	
Vienna University of Technology.	1. Pursuit	
http://www.ub.tuwien.ac.at/eng	profit	🛛 non-profit
	2. Organizati	onal sector
	🗌 public 🗌 p	private 🛛 non-governmental 🗌 other
	3. Operating	area
	🗌 local 🛛 🗌 r	national 🛛 international
	4. Working f	eld
	🗌 development a	aid 🛛 🖂 humanitarian aid 🗌 other
	5. Faith	
	🛛 faith-based	not faith-based
	6. Staff	
	volunteer-base	ed 🛛 salaried
	7. Providing	
	🛛 goods	Service

Given this specific context, Medair faces typical challenges of humanitarian NGOs, assessed in Section 3.3.3, and some specific challenges due to the organization's history and strategic choices. Many characteristics and the deriving challenges are interrelated. The following will explain Medair's challenges.

5.2.1 General Challenges for Medair

1. Staff Turnover: As other humanitarian NGOs, Medair faces this challenge and has to spend a lot of effort to recruit, introduce, and train new staff every year. Medair identifies reducing staff turnover as a strategic objective for the next years.

2. Unpredictability: Medair's strategic decision to "go the extra mile" to reach the most remote communities and operate in the most challenging environments brings additional challenges for the organization. Medair runs programmes in Somalia and South Sudan, which are ranked as the two most corrupt countries on the Global Corruption Index (Transparency International, 2016). South Sudan is Medair's largest country programme. Looting and ongoing conflicts are issues Medair has to face in these countries. This unstable environment causes enormous psychological stress for employees and makes operational planning very difficult. - Case Study: Medair -

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

TŲ UB

TU UB WIEN Universitätsbiblio

The approved original version of this diport of the version of the most remote places makes transportation and communication http://www.ub.tuwien.ac.at/eng very challenging. Information flows through the organization are interrupted and real time data is usually not available to Headquarters. For further reference, Rominger (2017) did a study on information flows for humanitarian aid operations.

> 4. Recruiting Staff with Organizational Fit: Medair is a faith-based organization and looks for people who fit into the organization's values of Christianity and are able to live in the challenging environments where Medair works. This means that Medair looks for committed Christians for their internationally recruited staff. The percentage of committed Christians in Europe, where Medair usually recruits, is declining (Stourton, 2015). According to Pew Research Center (2011), the percentage of weekly church attendees is 11% in Germany, 10% in France, and 9% in the United Kingdom. The percentage decreases when taking only the working age population into account. This means that Medair focuses on less than 10% of the already limited available human resources for humanitarian work. Medair does not permit internationally recruited staff to bring their families with them, and they do not wish to split up families. Therefore, internationally recruited staff cannot have children living at home, and if they are married, their spouse must also work for Medair. In addition to these restrictions, few people are willing to take the risks and endure the challenges of living and working in crisis contexts. Furthermore, few people are able to deal with the uncertainty and short (1-2 year) employment commitments of humanitarian work. All these factors together make it very challenging for Medair to find highly qualified internationally recruited staff.

> 5. Low use of Technology and Management Trends: The decision to only operate in humanitarian aid projects, makes it more difficult to get funding for investment in any technology or staff training. In development projects there is more flexibility to fund such investments (Parris, Thoughts on Lean in Humanitarian Aid, 2017).

6. Underspending and Earmarked Funding: Short spending deadlines and inflexible cost components can generate inefficiencies in humanitarian operations

- Case Study: Medair -

TU UB WIEN Universitätsbibli

The approved original version (IASC, 2016, p. 9). NGOs are under pressure to spend all earmarked funds due to mater thesis is available at the main library of the Vienna University of Technologa deadline, which does not contribute to effective spending of funds, and which http://www.ub.tuwien.ac.at/eng may even hinder cost-reduction improvements. As mentioned in Section 5.1.2, earmarked funding determines in which programme and budget line items the funds must be spent. Medair's percentage of earmarked funding is higher than the sector average which makes Medair's financials not flexible enough in this dynamic environment. In addition, some institutional donors required NGOs to raise a percentage of private funding in order to provide their own funding for a programme. As for many NGOs, raising enough private funds is a challenge for Medair.

7. Short Funding Cycles: Humanitarian operations are meant to reduce suffering during and after crises. Therefore, the time frame is short and usually about 12 months. Still, protracted crises exist, where communities need assistance for a longer period. The funding for humanitarian aid is still based on a 12-month cycle for less. To fund country programmes for another year, NGOs have to make a huge effort to attract and maintain funding. This consumes significant energy and time. Organization for Economic Co-operation and Development (OECD) has realized this problem and tries to shift donors from annual to a multi-annual funding cycles (OECD, 2017, p. 2ff), but the common practice is still 6-12 months.

8. Headquarters in Switzerland: The HQ located close to the UN in Geneva has advantages for humanitarian NGOs, but is also a challenge for overhead costs, due to the high salaries of Switzerland which has the worldwide fourth highest gross domestic product per capita (GDP) after Liechtenstein, Monaco, and Luxemburg which are considerably smaller countries (UN, 2015). Employees at headquarters are more expensive for Medair than in other countries.

5.2.2 Faith-Based Organization and Lean

Section 3.3.3 assesses the applicability of lean for humanitarian NGOs and Section 1.3, assessed the need of lean for humanitarian NGOs. All of these assessed needs or challenges for lean hold true for Medair, being a humanitarian INGO. One specific factor, namely being a faith-based organization is found to promote lean. Lean aligns well with

UB Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

- Case Study: Medair -

the Bible and Christianity in general (Parris, Thoughts on Lean in Humanitarian Aid, master thesis is available at the main library of the Vienna University 2017): If someone truly believes in something and is able to make the link between lean http://www.ub.twien.ac.at/eng and his or her beliefs, this may help to internalize lean easier and appreciate it more.

> Considering that Medair is in general a humanitarian NGO operating in more remote places, which puts even more pressure on operations and staff, the need for having optimal processes in place is even greater. This leads to an even greater need of lean for Medair. This means that Medair has and greater need of lean and a possible advantage to implement lean, if the connection to Christian faith can be shown to the employees.

TŲ UB Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Tech-nischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

TU UB

The approved original velot **Meddalit** is Lean Roadmap Vienna University of Technology.

In 2016, Medair's top level management realized that the organization has problems with their bureaucracy and processes, due to their rapid growth over the last years. In Medair's 2017-2019 strategy, they identified operations as a vulnerability and set Strategic Objective 7 - Operational Excellence and Leadership: "Build a healthy, agile, and sustainable organization that is known for excellent leadership and rigorous implementation." Process improvement and cultivating a culture of continuous improvement are initiatives within this objective. Medair wants to grow further and the plan is to optimize processes and make work easier so that the organization's capacity grows without adding costs. Basically, Medair wants to increase their performance and efficiency to be able to reach more people in need and deliver higher quality to beneficiaries. Therefore, lean can be the methodology to achieve this outcome.

6.1 **Process Excellence Manager**

In order to become a lean organization, Medair created a position called Process Excellence Manager (PEM) and recruited Andrew Parris, a Lean Six Sigma Master Black Belt with NGO experience, to fill the position.

6.1.1 The Person

Andrew Parris is a process improvement expert with over 20 years of experience and a PhD degree in Mechanical Engineering from MIT. He earned his PhD as part of the Lean Aerospace Initiative (LAI) which came out after the success of the IMVP. After finishing his PhD, Andrew Parris worked for Lockheed Martin as Process Improvement Lead for 11 years. In this time, he got certified as a Lean Six Sigma Black Belt in 2003. In 2008, he changed industries and moved to the development and humanitarian NGO sector, to join World Vision International, where he worked for nearly nine years. Three years of his time at WVI, he lived in East Africa and achieved remarkable results for WVI East Africa: \$1.5 Mio cost cut in recurring costs, and reducing lead time of recruitment and procurement by 40-80% (Parris, Improving Processes for Good in East Africa, 2013, p.

- Medair`s Lean Roadmap -

TU UB

The approved original version in January 2017, he started to work for Medair, as Process Excellence Manager, master thesis is available at the main library of the Vienna University in the Ecublens Headquarters.

6.1.2 The Role

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TŲ UB

As shown in Figure 5-4, the PEM reports to the Executive Officer, and is a manager that has no direct team to manage. The task of the PEM is to assist all departments in improving their processes. To achieve this, the PEM has three main functions:

1. Build improvement capabilities of employees to enable them to improve their own processes. This is done by training, facilitating workshops and coaching employees.

2. Provide expertise and guidance to employees during their improvement projects and in the application of lean concepts and tools. The PEM is not an expert in specific departments, but an expert in improving processes and making them lean. Usually teams work on process improvement projects, so the PEM coaches a process mapping session for instance.

3. Design and manage the overall lean transformation initiative so that it achieves strategic change and measured impact. This means getting everyone from senior leadership to the front lines field staff actively involved, making improvements, and sharing impact and learning.

The PEM and the Executive Office Director recently established and will lead the Medair Project Management Office (PMO). This creates a synergy, because all projects are about improvement, and all the process improvement projects need to be overseen by the PEM anyway. Although the negative effect is that the PMO work draws time and attention away from the lean initiative.

The PEM was assigned to a third task. Medair wants to get certified by the Core Humanitarian Standard on Quality and Accountability. To achieve this, the organization has revise or create new policies, systems and processes to meet the standard. The CHS draws together nine key elements of existing humanitarian standards, as shown in Section 4.3.1. Number seven states that humanitarian actors commit to continuously improve and learn (Core Humanitarian Standard, 2014, p. 9). This aligns perfectly with



- Medair's Lean Roadmap -

The approved original wedair, s plan to become a lean organization and makes sense that the PEM owns this master thesis is available at the main library of the Vienna University task, but again draws away time and attention from the actual task the PEM was hired http://www.ub.tuwien.ac.at/eng for, the lean implementation.

6.2 The Lean Roll-Out

To transform Medair into a lean organization, the lean initiative will be first rolled out at Medair HQ and in later stages will be brought to the country programmes. The first stage of the initiative is to build awareness of what lean, continuous improvement, and operational excellence means as shown in Figure 6-1: Medair's Lean Roll-out under Stage 1. The completion of how the adapted version of lean, called Process Excellence, is defined by how the processes should be, which is shown in the Appendix.



Figure 6-1: Medair`s Lean Roll-out

- Medair`s Lean Roadmap -

TU UB WIEN Universitätsbibli

The approved origined 12.1 of this diperfecteess Excellence Energizer Training (Parris, Process master thesis is available at the main library of the Vienna University Dechnological (Parris, 2017)

Stage 1 of the lean roll-out, is achieved through training staff in the Process Excellence Energizer Training. It introduces participants to process improvement and lean in the humanitarian NGO context. It is a one-day training that starts with a hands-on simulation in the morning that shows the difference between a bad and an excellent process. The afternoon contains of teaching process improvement tool sets such as visual management, 5S, and the 8 ways of waste. After the training the participants should know about the basic concepts of lean, process improvement, and should be able to apply the tools taught in the training.

By the end of 2017, over 80% Medair Headquarters staff should have gone through this training to ensure that all members know and are able to apply the basics of process improvement. The training creates awareness of the lean initiative and is the first step to equip Medair's staff with process improvement capabilities and to begin to build a culture of continuous improvement.

6.2.2 Lean Six Sigma Green Belt Training

There is no one globally agreed on standard on lean six sigma certification, but some common understandings of different skill levels. The scale is taken from martial arts, and one can commonly get certified at four (ASQ, 2017):

- 1. Lean Six Sigma Yellow Belt
- 2. Lean Six Sigma Green Belt
- 3. Lean Six Sigma Black Belt
- 4. Lean Six Sigma Master Black Belt

Medair is training and certifying Green Belts in the second stage of the lean initiative. Andrew Parris, as certified Master Black Belt, has the authority to certify Green Belts.

The International Association for Six Sigma Certification (IASSC) defines a certified Lean Six Sigma Green Belt as: "a professional who is well versed in the core to advanced
- Medair`s Lean Roadmap -

TU UB Universitätsbiblio

The approved original elements of Lean Six Sigma Methodology, who leads improvement projects [...] A Lean Master thesis is available at the main library of the Vienna University Six Sigma Green Belt possesses a thorough understanding of all aspects of the Lean Six Sigma Method including competence in subject matters contained within the phases of Define, Measure, Analyze, Improve and Control (DMAIC) [...] A Lean Six Sigma Green Belt understands how to implement, perform, interpret and apply Lean Six Sigma at a high level of proficiency." (2017)

To become a certified Green Belt, a participant has to fulfil three different requirements.

- 1. Attend all training sessions
- 2. Pass the final Green Belt exam
- 3. Complete the Green Belt Project

The aim of the training is to build Change Agents in the organization. After being certified, these Green Belts should drive and carry the lean initiative in the organization, improve processes, and help their colleagues to improve processes providing their expertise gained through the training.

6.2.3 Building Lean Leaders

As mentioned in Section 4.2.1, to successfully implement a lean culture, management is the single most important determinant. Therefore, Medair has to build lean leadership capabilities in the top level management. The plan to achieve this is still being developed. Before the lean initiative started, Medair already planned to develop its leadership style and training, not considering lean at this point. Now the question raises of how the agreed leadership style aligns with being a lean leader. To become a truly lean organization the leadership has to commit to the lean leadership style. This is a major element in the lean implementation for Medair.

6.2.4 Deploy Lean to the Field

All operations of Medair run in different country programmes, and this means Medair has to bring lean initiative to each country programme. To deploy the initiative to the field, generally, it is very difficult, as mentioned in Section 3.2, to get funding for overhead expense which this would definitely be. Another challenge is to find the right people and

TU UB WIEN Universitätsbibliothek

- Medair`s Lean Roadmap -

The approved origin place to train staff. Staff turnover in the field is very high and it does not help the viena University organization to build capacity in people that leave soon. Medair operates in very remote and dangerous places, the work units are usually very small. It is necessary to get people of different country programmes to one easy to reach location to facilitate training there. This assures that many employees get training, but adds travel costs and additional time away from their valuable work.



The approved original vor **Review** to **Medair`s Approach** master thesis is available **T** the **Review** to **Medair`s Approach** Vienna University of Technology.

This chapter assesses how successful the lean roll out at Medair has been until 20th of August 2017. The Process Excellence Manager started to work for Medair in the Beginning of 2017. The time frame of the assessment is therefore 33 weeks. As mentioned above, a lean transformation takes about 10 years (Zarbo, 2012, p. 321). This study has a considerably short time frame to observe an organizational cultural change. Still, there are already some interesting insights about the kick-off and early phase of the implementation. The following sections are in the order of the lean roll-out shown in Figure 6-1.

7.1 Stage 1: Process Excellence Energizer Training

Until mid of August 2017 the PEM trained already 85 Medair employees, and an additional 10 external participants from various institutions, such as ETH Zurich, Kühne Help Logistics, and others. 57 out of the 85 trained Medair staff work in Headquarters. At headquarters there around 110 employees, meaning that in 33 weeks already half of the staff at Headquarters have participated in the one-day training, the content explained of the training is explained in Section 6.2.1. This means that stage 1 of the initiative, building awareness and small improvement skills in staff, can be already seen as successful from a training at headquarter ratio. Training skills, however, does not automatically mean that people use them and that Medair benefits from it.

One event that may shows how staff that has participated in the training, is likely to use the trained skills, is the I/We-Made-it-Better challenge. It is an opportunity to show other employees in Medair how someone or a team made a small, documented improvement using the improvement methodology learned in the Process Excellence Energizer Training or Green Belt Training. A deadline was set to upload the documentation of improvements, but it was not mandatory for employees to upload an improvement. The deadline was in mid-August 2017; some people were on vacation which may lower the amounts of improvements made, shown in Table 7-1. Out of the 57 trained staff, 34 improvements were uploaded, and the table also shows how many employees in every department were trained and uploaded their improvements.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

The approved original version of this diploma or

http://www.ub.tuwien.ac.at

http://www.ub.tuwien.ac.

TU UB WIEN Universitätsbibliot - Review of Medair's Approach -

Table 7-1: I-We Made it Better (WIMB)-Improvements

blogy. t/eng Department	Number Who Took Full PE Energiser	Number of WIMB Improvements	% with documented I-We Made It Better
Finance	11	9	82%
Human Resources	9	5	56%
International			
Programs	10	2	22%
Information Systems	5	4	80%
Logistics	7	8	100%
Marketing &			
Relationships	9	5	56%
Executive Leadership			
Team	6	1	17%
Total	57	34	60%

In total 60% of the employees who were asked to do improvement, did so. This is not a high percentage, but it does show that small improvements are being made and documented. The most problematic is probably that only one of the executive leader team has uploaded an improvement. Part of the reason for this low number is that the work of executives is typically strategic in nature, rather than operational, so that they have much less opportunity to make small improvements. It could also be seen as an indicator that the leadership buy-in of the initiative has not yet fully developed. Everyone at Medair can see the documented improvements and who did them. If people, see that only one of their executive officers did improvements they might think that the lean initiative is not recognized as much by the ELT. This may discourage staff from spending effort on improvements, if they feel it is not appreciated by their bosses. The building of lean leaders comes at stage 2 of the initiative, but it already shows how important leaders are in the initial stage.

7.2 Stage 2: Developing Change Agents and Lean Leaders

The first five-day Green Belt Training took place between March and April 2017. It was held by Andrew Parris, the Process Excellence manager, as a five one-day training sessions spread over six weeks. 19 Medair employees started the training, two had to drop out during the training, because of various reasons. The 17 remaining work in

TŲ UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich. http://www.ub.tuwien.ac.at

- Review of Medair's Approach -

The approved original departments of the Headquarter and hold different position. Table 7-2 gives a master thesis is available at the main library of the Vienna University closer view of the participants.

Department	Number of participants	Manager / Team Member	
Information Service	2	1/1	
Programmes	1	1/0	
Human Resources	5	1/4	
Finance	4	0/4	
Marketing &	2	0/2	
Relationships			
Logistics	3	0/3	

Tahla	7_2.	Green	Rolt	Training	Particir	nants
rabic	1-2.	Green	DUIL	riannig	i articip	Junio

Before participating in the training every participant had to identify one process they want to improve in their work area, and get commitment from their manager and director that they will work on this project through to completion. The Green Belt Project has three main purposes: Apply the learning, build skills, and show colleagues the benefits of lean.

12 different improvement projects were identified, and only one has been finished. Staff are very busy with their everyday work, so that the Green Belt projects take longer than expected, which was three months. Out of the training two participants have finished their projects and got certified. One left Medair shortly after finishing and one was extern. So, no participants in Medair are certified Green Belts yet. Still, all participants went through the whole training and developed problem solving and improvement skills, and are now applying these in their Green Belt projects and in their everyday work. They are also acting as change agents.

The development of lean leaders, meaning managers and executive leaders, has yet not taken place, except for the three managers who participated in the Green Belt training.

7.3 Stage 3: Improvement Project

The first part of this stage has already been made with the WIMB projects, shown in Section 7.1. Next, the PEM is planning to do Kaizen workshops of two or three days in HQ in the departmental work groups to improve key processes in the respective department.

- Review of Medair's Approach -

TU UB WIEN Universitätsbibliothe

approved origin 7.4 Stage 4: Lean Daily Meetings and Visual Boards

master thesis is available at the ma Vienna University of Technology.

http://www.ub.tu Thisatpart has not been rolled out yet, except for a team in the information services department. (The manager of this work group – one of the Green Belts – already held daily stand-up meetings before the lean initiative.) However, two Green Belts plan to start this practice in their areas.

7.5 Bringing Lean to the Field

Bringing Lean to the field would basically follow the same approach that is being taken in Medair headquarters in Ecublens, Switzerland, described above.

To realize field training, there is a proposal to cooperate with Kühne HELP Logistics, SpreadThinking, and ETH Zürich. HELP Logistics is part of the Kühne Foundation and offers consulting and training in humanitarian supply chain management (Kühne-Stiftung, 2017). SpreadThinking is a non-profit organization with the purpose of spreading continuous improvement philosophy and principles (SpreadThinking, 2017). This collaboration wants to offer a 4-day workshop to teach about continuous improvement and lean in the humanitarian NGO context. This workshop is not meant for only Medair, but also for other NGOs to participate. The idea of this workshop was born in June 2017 and the details are not clear yet, but it is a huge opportunity to spread lean into the humanitarian sector.

Medair is also considering the idea of partnering with a Swiss corporation that has benefitted from applying Lean and that would be eager to share its Lean expertise with Medair as part of its corporate social responsibility. Given the importance and visibility of the Middle Eastern refugee crisis in Europe, this would provide a corporation the opportunity to help displaced people in the Middle East, by helping Medair improve its processes, both in Switzerland and in the field.



This thesis now concludes by answering the three research questions, stating its limitations, and providing recommendations for Medair in their lean transformation journey.

8.1 Answer Research Question 1: Can Lean be applied to benefit a humanitarian NGO?

Yes.

As shown in the assessment in Section 3.3.3, there are serious impediments to establishing lean in humanitarian NGOs. The contextual differences to the business sector overall make it more difficult to implement lean in the humanitarian sector. The most critical factor is found to be the high staff turnover rate. The assessment also shows that lean cannot be applied to the same extent as for manufacturing companies and has to be adapted to benefit humanitarian NGOs best. Section 4.1 shows that lean is applicable to the context and beneficial for humanitarian NGOs.

As the above parts of this paper answer that lean can benefit a humanitarian NGO, Section 4.3 identifies that there is an actual and great need for lean in the humanitarian aid sector. Lean is needed to enable humanitarian NGOs to deliver better services and be able to respond faster and better to disasters. This will enable them to save more lives, relieve more suffering, and promote stronger recovery.

8.2 Answer Research Question 2: If so, what aspects of Lean are most useful?

The book The Routledge Companion to Lean Management shows that it is possible to take lean or lean tools out of manufacturing and apply them to other sectors successfully. The application areas of lean seem to be unrestricted, but to be successful, lean needs to be adapted to fit to the context. Creativity and wisdom in the application and adaption of lean tools is important to achieve good results.

- Conclusions -

TU UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TŲ UB

The approved original tool applicability assessment, Section 4.1, shows that many lean tools are interest is available at the main library of the viena University applicable in humanitarian aid and the literature review proves this by giving some examples of successful lean implementations in the humanitarian aid and the non-profit sector. 8 of 13 important lean tools are identified to be applicable to humanitarian aid and the other five partly applicable. As people develop their improvement skills, they may find more creative ways to apply lean tools to benefit humanitarian operations. This review confirms findings of Rüttimann, et al. (2014, p.359) who say that adapted lean tools offer a huge improvement potential benefit for the service industry. Humanitarian NGOs provide services and can be therefore seen as a part of the service industry.

The applicability of lean thinking is found to be beneficial for any organization, regardless the sector. An organizational culture that promotes and lives continuous improvement automatically increases the performance of the organization by solving problems systematically, developing sustainable solutions, and continuously improving work processes.

8.3 Answer Research Question 3: How has it been deployed in Medair and other humanitarian NGOs?

The literature review and the tool assessment show that most lean tools are applicable and some are already successfully implemented in the humanitarian aid context. This means that lean tools are applicable to benefit humanitarian operations. Although not many humanitarian NGOs have thought to apply lean so far, and tool applications are limited, there is a high potential benefit for these organization.

The literature review in Section 4.1.1 shows that lean tools have been applied to humanitarian NGOs and showed success. WVI went one step further and implemented not only lean tools, but tried to build a lean culture in their East Africa section which achieved great results. Medair is the first humanitarian NGO that tries to implement lean organization-wide. As shown in Chapters 1 and 7, lean has been adapted to the humanitarian NGO context and is there called 'Process Excellence', which is an adaption of mostly lean and some parts of six sigma. The organization`s size makes it possible to implement lean throughout the whole organization with one expert. The first step of

TU UB

- Conclusions -

The approved original version of the version of the

The general idea is to develop problem solving skills in the employees and practice to improve the organization's performance. This means teaching employees lean thinking and lean tools, and expecting them to apply them to improve processes, which begins to establish a lean culture (Parris, Thoughts on Lean in Humanitarian Aid, 2017), as discussed in Section 4.2.1. Although it uses the term different term 'Process Excellence', Medair still aims to develop a lean culture.

The author of this thesis had the opportunity to volunteer as assistant of the Process Excellence Manager, Andrew Parris, during the duration of this project. During the time at Medair, the author experienced Medair's lean roll-out first hand and participated in the Green Belt training and Process Excellence Energizer Training. From personal experience the author can say that the initial stage of the lean implementation at Medair can be considered successful. The majority of employees was open to this initiative and welcomed the opportunity to learn about lean and process excellence. The WIMB challenge showed that people are already using the methods learned in the trainings. The initial stage was a success for Medair, but only the future will show if the organization will complete the full lean transformation.

8.3.1 Limitations

Due to the limited observation time of 5 months, only the early stages of the lean transformation at Medair could be observed and discussed in this paper. The future will show how Medair can overcome the sectorial obstacles of a successful lean transformation, identified in Section 3.3.3, such as funding issues and employee turnover. It will take years to be able to say if their lean transformation was successful. Further observation of this lean transformation is interesting for academic research, because it is the first organization-wide lean implementation with a permanently employed expert at a humanitarian NGO.

- Conclusions -

http://www.ub.tuwien.ac.at

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

TŲ UB

The author volunt eered at Medair and had no experience in humanitarian aid before. The approved original the main library of the Vienna University Therefore, the objectivity of the assessment may be biased towards the experiences at Medair and the lack of reference of other NGOs. Still, the effort was to use many reliable sources to make the most possible objective assessment.

8.4 Recommendations for Medair

From the beginning of the Green Belt training the author experienced that people were welcoming to the idea of learning skills that will help them to improve their everyday work life. After the training, a group feeling among the Green Belt participants established that is very beneficial for the initiative and motivational for all members. The first steps of the initiative were successful, but to make the whole initiative a successful there is still a long way to go. The author gives the following suggestions to Medair which should help to transform successfully:

• Prioritize Lean or Process Excellence to be successful: Medair appears understaffed in all areas and people have to work very hard. So, it is hard for employees to see the benefit of stopping their usual work and improve processes to ease their work. From the author's experience at Medair, lean has not been prioritized yet which slows down the transformation.

• Incentivize process improvement: Continuous work on process improvement has to be part of the work for every employee. It will be helpful to put process improvements in the annual employee performance assessment.

• Build lean leaders: There are existing leadership styles that Medair has agreed on before starting the lean initiative. The problem is now that lean comes along with a special leadership style. Now this lean leadership style has to be adopted to Medair. Leadership has been found to be the most important determinant of success in a lean transformation (Byrne, 2017 & Zarbo, 2012), as shown in Section 4.2.2. Medair already made a big step forward by committing to this initiative, but to make it successful the executive leaders and managers have to lead it. This has not yet been realized by all leaders. Table 4-2 shows how lean leaders differ from usual leaders. Developing lean leaders is of vital importance to

- Conclusions -

The approved original version de this optimation and leaders have to start internalizing lean thinking and be taught master thesis is available at the main library of the Vienna University of Technologhow to act as a lean leader.

• Organizational Structure: As shown in Figure 5-4, Medair has a clear functional structure in their Swiss HQ. A functional structure promotes employees to become specialists in their field (Gupta, 2009). Developing skills and knowledge requires time which is in conflict with the high staff turnover. Lean promotes generalists and team work and therefore works better in a cross-functional structure. One example of how this could cross-functional structure could look, is shown in Figure 8-1.

Because specialization is lower in cross-functional teams, it is easier to replace staff and responsibilities can be shared, so if a person leaves, it is easier for a new employee to step in. Silo thinking and bureaucracy is also reduced in this system (Gupta, 2009). This is in some way done at Medair, usually one person is specialized on one or two programmes, but employees` workplace is located in their functional department, for instance finance or logistics. This creates silo thinking and hinders communication throughout departments. This structure change means a major change for the organization, but has the chance to increase Medair`s performance and the success of the lean initiative.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



Figure 8-1: Possible Cross-functional Structure at Medair

TŲ UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at



The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng

Appendix

Process Excellence

Effective: Our processes are reliable and provide relief

- Reliable: They consistently and safely deliver quality, valued outputs and impact
- Relief: They relieve or facilitate relieving human suffering for the most vulnerable communities

Efficient: Our processes are simple, standard, timely and integrated

- Simple: They are intuitive, clear, concise, visually organised, and easy to perform correctly
- Standard: They use defined, common workflows, tools, roles & responsibilities, policies, and expectations
- Timely: They flow smoothly (minimum hand-offs, delays, rework) to provide output when needed
- Integrated: They seamlessly integrate with other processes and into a larger, optimised system

Appropriate: We respect all stakeholders, are flexible, and use appropriate technology

- **Respect**: We honour all who are involved in or affected by our work, and their rights, values and beliefs
- Flexibility: We adapt to different contexts and respond easily to changing situations
- Technology: We use reliable technology that is appropriate to the purpose, user, and use environment

Empowering: We make decisions locally, partner with stakeholders, and have needed capacity

- Local Decisions: We make decisions transparently and as close to the action as possible
- · Partnering: We work with stakeholders to increase capacity, teamwork, outputs, and impact
- · Capacity: We have the people, tools, skills, knowledge, and work culture we need to achieve outputs and impact

Continuously Improving: We are accountable, correct problems quickly, and apply learning

- · Accountability: We own, measure, report, review, and act on process performance, outputs, impact and feedback
- Correcting Problems Quickly: We make problems visible, promptly investigate them, and address root causes
- · Learning: We reflect, and proactively develop, apply, and share learning, best practices, improvements and innovation

MEDAIR 113

Appendix A: Process Excellence 1

The approved original version of this diploma or master thesis is available at the main library of the Vienna University of Technology. http://www.ub.tuwien.ac.at/eng



IASC. (2016). <i>Donor Conditions and Their Implication for Humanitarian Response</i> . Inter-Agency Standing Committee.
Abad, E. T. (2013, December 5). <i>slideshare.net</i> . Retrieved May 29, 2017, from https://www.slideshare.net/abad_er/public-administration-versus-private-
administration
Akers, P. A. (2016). 2 Second Lean (7 ed.). USA: FastCap Press.
Andersson, R., Eriksson, H., & Torstensson, H. (2006). Similarities and differences between TQM, six sigma and lean. <i>The TQM Magazine</i> , <i>18</i> (3), 282-296.
ASQ. (2017). ASQ. Retrieved July 5, 2017, from https://asq.org/training/catalog
Balle, M. (2009, May 26). <i>http://michaelballe.fr.</i> Retrieved August 23, 2017, from http://michaelballe.fr:
http://michaelballe.fr/papers/LeanLeadershipPeopleDevCapeTown.pdf
Baudin, M. (2013, January 9). http://michelbaudin.com. Retrieved June 12, 2013, from
http://michelbaudin.com/2013/01/09/mit-article-comparing-lean-tqm-six-sigma-and- related-enterprise-process-improvement-methods/#methods_and_descriptions
Beamon, B. M., & Balcik, B. (2008). Performance measurement in humanitarian relief chains.
International Journal of Public Sector Management , 21(1), 4-25.
Billis, D. (2010). <i>Hybrid Organizations and the Third Sector</i> (1st Edition ed.). Hampshire: Palgrave Macmillan.
Bowen, D. E., & Youngdahl, W. E. (1998). "Lean" service: in defense of a production-line approach. International Journal of Service Industry Management, 9(3), 207-225.
Branczik, A. (2004, February). <i>http://www.beyondintractability.org.</i> (G. Burgess, & H. Burgess, Eds.) Retrieved May 20, 2017, from
http://www.beyondintractability.org/essay/humanitarian-aid
Bray, J. D. (2001). Developing Mitigation Measures for the Hazards Associated with Earthquake
Surface Fault Rupture. Tokyo: Japan Society for the Promotion of Science.
Byrne, A. (2017). <i>The lean turnaround action guide: how to implement lean throughout.</i> New York: McGraw-Hill Education.
Chandra, P. V. (2013, June 25). <i>12 Essential Lean concepts and tools</i> . Retrieved July 26, 2017, from www.processexcellencenetwork.com:
https://www.processexcellencenetwork.com/lean-six-sigma-business-
transformation/articles/12-essential-lean-concepts-and-tools
Charity Navigator. (2017). <i>charitynavigator.org</i> . Retrieved July 26, 2017, from charitynavigator.org:
https://www.charitynavigator.org/index.cfm?bay=content.view&cpid=48#FinancialEffici encyPerformanceMetrics
Chen, H. (2011). Research on uncertainty demand inventory control in supply chain. 8. Harbin: IEEE.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbibli

The approved original chiarini, A. (2011), Japanese total quality control, TQM, Deming's system of profound master thesis is available at the main broker, BPR, Lean and Six Sigma. *International Journal of Lean Six Sigma, 2*(4), 332http://www.ub.tuwien.ac.at/e355.

- CHS Alliance. (2016, July). *chsalliance.org*. Retrieved August 14, 2017, from http://www.chsalliance.org/what-we-do/chs
- Clark, J. (1995). The state, popular participation, and the voluntary sector. *World Development, 23*(4), 593-601.
- Collins, J. C., & Porras, J. I. (1996). Built to Last. Century Business.
- Core Humanitarian Standard. (2014). *Core Humanitarian Standard on Quality and Accountability*. CHS Alliance, Groupe URD and the Sphere Project.
- Cozzolino, A., Rossi, S., & Conforti, A. (2012). Agile and lean principles in the humanitarian supply chain: The case of the United Nations World Food Programme. *Journal of Humanitarian Logistics and Supply Chain Management*, 2(1), 16-33.

DARA. (2013). Now or Never- Making Humanitarian Aid More Effective. Spain: DARA.

- Davoren, J. (2016, June 23). http://smallbusiness.chron.com. Retrieved June 6, 2017, from http://smallbusiness.chron.com/functional-structure-organization-strength-weakness-60111.html
- Development Initiatives. (2016). *Global Humanitarian Assistance Report 2016.* Development Initiatives.
- Dombrowski, U., & Mielke, T. (2013). Lean Leadership fundamental principles and their application. *Procedia CIRP*, 7, 565-570.
- Drucker, P. F. (1996). *Managing the Nonprofit Organization (Principles and Practices).* UK: HarperCollins Publisher.
- Dubey, R., Gunasekaran, A., Altay, N., Childe, S. J., & Papadopoulos, T. (2016). Understanding employee turnover in humanitarian organizations. *Industrial and Commercial Training*, 48(4), 208-214.
- EL-Naggar, M. (2013, July 23). In Lieu of Money, Toyota Donates Efficiency to New York Charity. *New York Times*.
- English Oxford Living Dicitonaries. (2017). *English Oxford Living Dicitonaries*. Retrieved May 24, 2017, from https://en.oxforddictionaries.com/definition/ngo
- Eriksson, C. B., Bjorck, J. P., Larson, L. C., Walling, S. M., Trice, G. A., Fawcett, J., . . . Foy, D. W. (2009). Social support, organisational support, and religious support in relation to burnout in expatriate humanitarian aid workers. *Mental Health, Religion & Culture, 12*(7), 671-686.
- Ferris, E. (2007, October 24). Addressing the Gap between Relief and Development. Brookings.

Gannon, J. (2014, July). International Philanthropy and Disasters in Developed Countries: The US Response to Japan's 3.11 Disaster. Japan Center For International Exchange.

- Goldensohn, D. (2015, May 12). *All About Lean.* Retrieved August 24, 2017, from www.allaboutlean.com: http://www.allaboutlean.com/definition-of-lean/
- GRIPS Development Forum. (2009). *Introducing KAIZEN in Africa*. Japan: GRIPS Development Forum.
- Grunden, N. (2013). 55 Yields Unexpected Benefits at Global Links . Lean4NGO.

Gupta, A. (2009). Functional vs. Divisional Structure. practical-management.com.

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac

TU UB

The approved original Hartmann, B_p (2015, September 23). *Go Lean: How To Eliminate Waste and Increase Efficiency* master thesis is available at the name of the Farm. Retrieved July 7, 2017, from https://www.chelseagreen.com/blogs/gohttp://www.ub.tuwien.ac.at/elean-eliminate-waste-increase-efficiency-farm/

- Helten, K., & Lindemann, U. (2012). Structural Modelling and Analysis of Organizational Change during Lean Development Implementation. Dubrovnik: International Design Conference. Henry, J. (2004). Understanding HR in the Humanitarian Sector (1st Edition ed.). People in Aid.
- Holmemo, M. D.-Q., Rolfsen, M., & Ingvaldsen, J. A. (2016). Lean thinking: outside-in, bottom-up? The paradox of contemporary soft lean and consultant-driven lean implementation. *Total Quality Management & Business Excellence, 0*(0), 1-13.
- HQAI. (2017, August 14). HQAI. Retrieved August 2017, from hqai.org: http://hqai.org/verification/
- IASSC. (2017). *IASSC*. Retrieved July 5, 2017, from http://www.iassc.org/six-sigma-certification/green-belt-certification/
- ISO 9001 Checklist. (2015). *ISO 9001 Checklist Homepage*. Retrieved August 11, 2017, from https://www.iso-9001-checklist.co.uk/tutorial/8.5.1-continual-improvement.htm
- Joseph, C. (2017). *Smallbusinesses Chron.* Retrieved August 3, 2017, from http://smallbusiness.chron.com/benefits-disadvantages-functional-organizationalstructure-11944.html
- Kühne-Stiftung. (2017). *Kühne-Stiftung*. Retrieved July 6, 2017, from https://www.kuehnestiftung.org/index.php?id=10104
- Korff, V. P. (2012). *Between cause and control management in a humanitarian organization.* Groningen: University of Groningen.
- Korff, V. P., Balbo, N., Mills, M., Heyse, L., & Wittek, R. (2015, July). The impact of humanitarian context conditions and individual characteristics on aid worker retention. *Disasters*, 39(3), 522-545.

Kovács, G., & Spens, K. M. (2007). Humanitarian logistics in disaster relief operations. International Journal of Physical Distribution & Logistics Management, 37(2), 99-114.

- Legnerova, K. (2016). *Motivation in Profit and Non-Profit Organizations*. Prague: University of Economics in Prague.
- Lewis, D. (2009). Nongovernmental Organiztions, Definition and History. In H. K. Anheier, & S. Toepler (Eds.), *International Encyclopedia of Civil Society* (pp. 1056-1062). Springer US.
- Liker, J. (2003). *The Toyota Way: 14 Management Principles from the World's Greatest Manufacturer* (1st Edition ed.). New York: McGraw Hill Professional.
- Liker, J. K. (2016). The Routledge Companion to Lean Management. In T. H. Netland, & D. J. Powell (Eds.). Routledge.
- Liker, J., & Convis, G. L. (2011). *The Toyota Way to Lean Leadership* (1st Edition ed.). United States: McGraw Hill Professional.
- Liker, J., & Hoseus, M. (2008). *Toyota Culture: The Heart and Soul of the Toyota Way* (1st Edition ed.). United States: McGraw-Hill Education.
- LinkedIn. (2017, June 12).
- Liu, Y. (2009). Analysis and Evaluation of Organizational Change Approaches. *International Journal of Business and Management, 4*(12), 234-238.
- Loquercio, D., Hammersley, M., & Emmens, B. (2006, June). Understanding and addressing staff turnover in humanitarian agencies. *Humanitarian Practice Network, 55*, 1-30.

TU UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac

The approved origin Martens, K. (2002). Mission Impossible? Defining Nongovernmental Organizations. *Voluntas:* Wienna University of Technology *International Journal of Voluntary and Nonprofit Organizations, 13*(3), 271-285. http://www.ub.tu/McCormack, L., & Joseph, S. (2013). Psychological growth in humanitarian aid personnel:

reintegrating with family and community following exposure to war and genocide. *Community, Work & Family, 16*(2), 147-163.

Medair. (2016). Annual Report 2016. Ecublens: Medair.

Näslund, D. (2008). Lean, six sigma and lean sigma: fads or real process improvement methods? Business Process Management Journal, 14(3), 269-287.

Ndihokubwayo, D. (2015). Where Last Mile Mobile Solutions are inexistent, simple lean methodologies make general food distribution easy and quick. Sudan: Food for the Hungry.

Netland, T. (2014). Introduction. Lean Management Journal, 9(4), 6.

Netland, T. H., & Powell, D. J. (2017). Preface. In T. H. Netland, & D. J. Powell (Eds.), *The Routledge Companion to Lean Management* (1st Edition ed., p. XXI). New York: Routledge.

NIST. (2000). Principles of Lean Manufacturing with Live Simulation. Gaithersburg: NIST.

OECD. (2016). OECD.org. Retrieved June 14, 2017, from

http://www.oecd.org/dac/stats/humanitarian-assistance.htm

OECD. (2017). Multi-Year Humanitarian Funding. Paris: World Humanitarian Summit.

Ohno, T. (1988). *Toyota Production System: Beyond Large-Scale Production.* Oregon: Productivity Press.

Oloruntoba, R., & Gray, R. (2009). Customer service in emergency relief chains. *International Journal of Physical Distribution & Logistics Management*, *39*(6), 486-505.

- Ortiz, C. A. (2012). *The Psychology of Lean Improvements* (1st Edition ed.). Boca Raton: CRC Press.
- Osa, Y. (2012). *Growing Role of NGOs in Disaster Relief and Humanitarian Assistance in East Asia.* Japan Center For International Exchange.
- Parris, A. (2008). Green Belt Training. World Vision.

Parris, A. (2013). Improving Processes for Good in East Africa. *The TQM Journal, 25*(5), 458-472.

Parris, A. (2015). Fighting Poverty in East Africa with Lean Six Sigma. WVI.

Parris, A. (2017). Process Excellence Energizer. Ecublens: Medair.

Parris, A. (2017). Thoughts on Lean in Humanitarian Aid. Ecublens.

Peterkeni, H., & Bandara, W. (2015). *Business processes in international humanitarian aid.* Portail Qualite.

Pieńkowski, M. (2016). Implementation of Lean in Non-governmental Organizations. Wroclaw Food Bank Case Stuy. *International Journal of Contemporary Management*, *15*(1), 107-130.

Rüttimann, B. G., Fischer, U. P., & Stöckli, M. T. (2014). Leveraging Lean in the Office: Lean Office Needs a Novel and Differentiated Approach. *Journal of Service Science and Management*, 7, 352-360.

Rominger, A. (2017). *Information Flow in Humanitarian Relief Operations*. Zurich: Zürich School of Applied Sciences ZHAW.

Rother, M. (2009). Toyota Kata (1st Edition ed.). United States: McGraw-Hill.

TU UB

Die approbierte Originalversion dieser Diplom-/ Masterarbeit ist in der Hauptbibliothek der Technischen Universität Wien aufgestellt und zugänglich.

http://www.ub.tuwien.ac.at

TU UB WIEN Universitätsbibl

The approved original Smith, R., & Hawkins, B. (2004). Lean Maintenance: Reduce Costs, Improve Quality, and master thesis is available at the man bare (1st Edition ed.). Elsevier.

http://www.ub.tu/SpreadThinking. (2017). https://spreadthinking.org. Retrieved July 6, 2017, from

https://spreadthinking.org

Stamm, M. L., & Neitzert, T. (2008). *Key Performance Indicators for the Implementation of Lean Methodologies in a Manufacturing-to-Order Small and Medium Size Enterprise.* Auckland: AUT University.

Stourton, E. (2015). The decline of religion in the West. London: BBC.

Taylor, D. (2009). Evaluating Alternative Supply Chain Models for the Provision of Humanitarian Aid into Africa. (pp. 27-29). Cardiff: Humanitarian Development Summit.

- Thakur-Weigold, B., Stumpf, J., & Wagner, S. (2015, December). Helping the Helpers. *Industrial Engineer*, 47(12), 24-29.
- Tomasini, R. M., & Van Wassenhove, L. (2009). From preparedness to partnerships: case study research on humanitarian logistics. *International Transactions in Operational Research*, *16*, 549-559.
- Tomasini, R., & Van Wassenhove, L. W. (2004). A framework to unravel, prioritize and coordinate vulnerability and complexity factors affecting a humanitarian response operation, Working Paper No. 2004/41/TM. Fontainebleau: INSEAD.
- Toyota Global . (2017). *Toyota-global.com*. Retrieved July 20, 2017, from http://www.toyotaglobal.com/company/history_of_toyota/75years/data/automotive_business/production /production/japan/general_status/honsha.html
- Transparency International. (2016). *Corruption Perception Index 2016.* Transparency International.
- UN. (2015). *unstats.un.org.* Retrieved August 3, 2017, from https://unstats.un.org/unsd/snaama/selbasicFast.asp
- Van Wassenhove, L. N. (2006). Humanitarian aid logistics: supply chain management in high gear. *Journal of the Operational Research Society*, *57*(5), 475-489.

Wangwacharakul, P., Berglund, M., Harlin, U., & Gullander, P. (2014). Cultural Aspects when Implementing Lean Production and Lean Product Development. *Quality Innovation Prosperity*, 18(1), 125-140.

Warfield, C. (2008). The Disaster Management Cycle. Global Development Research Center.

- WebFinance. (2017, August 15). *BusinessDictionary.com*. Retrieved August 17, 2017, from http://www.businessdictionary.com/definition/customer.html
- Womack, J. P. (2006). Lean Thinking Works Everywhere! Jacksonville: Lean Enterprise Insititute.

Womack, J., & Jones, D. T. (1996). Lean Thinking (1st Edition ed.). New York: Simon & Schuster.

Womack, J., Jones, D. T., & Roos, D. (1990). The Machine That Changed the World. Free Press.

Zarbo, R. J. (2012). Creating and Sustaining a Lean Culture of Continuous Process Improvement. *Am J Clin Pathol, 138*(3), 321-326.